



# Article Microtomographic Analysis of a Palaeolithic Wooden Point from the Ljubljanica River<sup>†</sup>

Enej Guček Puhar <sup>1,\*</sup>, Lidija Korat <sup>2</sup>, Miran Erič <sup>3</sup>, Aleš Jaklič <sup>1</sup>, and Franc Solina <sup>1,\*</sup>

- <sup>1</sup> Computer Vision Laboratory, Faculty of Computer and Information Science, University of Ljubljana, Večna Pot 113, SI-1000 Ljubljana, Slovenia; ales.jaklic@fri.uni-lj.si
- <sup>2</sup> The Laboratory for Cements, Mortars and Ceramics, Slovenian National Building and Civil Engineering Institute, Dimičeva Ulica 12, SI-1000 Ljubljana, Slovenija; lidija.korat@zag.si
- <sup>3</sup> Institute for the Protection of Cultural Heritage of Slovenia, Poljanska 40, SI-1000 Ljubljana, Slovenia; miran.eric@guest.arnes.si
- \* Correspondence: enej.gucek.puhar@gmail.com (E.G.P.) ; franc.solina@fri.uni-lj.si (F.S.)
- + This paper is an extended version of our paper published in Guček Puhar, E.; Korat, L.; Erič, M.; Jaklič, A.; Solina, F. Reconstruction of 3D models from microtomographic images of archeological artifacts. In Proceedings of the 2020 IMEKO TC-4 International Conference on Metrology for Archaeology and Cultural Heritage, Trento, Italy, 22–24 October 2020.

Abstract: A rare and valuable Palaeolithic wooden point, presumably belonging to a hunting weapon, was found in the Ljubljanica River in Slovenia in 2008. In order to prevent complete decay, the waterlogged wooden artefact had to undergo conservation treatment, which usually involves some expected deformations of structure and shape. To investigate these changes, a series of surfacebased 3D models of the artefact were created before, during and after the conservation process. Unfortunately, the surface-based 3D models were not sufficient to understand the internal processes inside the wooden artefact (cracks, cavities, fractures). Since some of the surface-based 3D models were taken with a microtomographic scanner, we decided to create a volumetric 3D model from the available 2D tomographic images. In order to have complete control and greater flexibility in creating the volumetric 3D model than is the case with commercial software, we decided to implement our own algorithm. In fact, two algorithms were implemented for the construction of surface-based 3D models and for the construction of volumetric 3D models, using (1) unsegmented 2D images CT and (2) segmented 2D images CT. The results were positive in comparison with commercial software and new information was obtained about the actual state and causes of the deformation of the artefact. Such models could be a valuable aid in the selection of appropriate conservation and restoration methods and techniques in cultural heritage research.

**Keywords:** computer vision; computed tomography; 3D surface-based models; 3D volumetric models; Palaeolithic wooden point; conservation; waterlogged wood; archaeological documentation; heritage science

# 1. Introduction

Our interest in reconstructing 3D solid models using computed tomography was piqued when we were confronted with the deformation changes of a Palaeolithic wooden point [1] (Figure 1). The 40,000-year-old wooden point was found underwater in the Ljubljanica River close to Sinja Gorica (Figure 2) in 2008 [2]. The Ljubljanica River flows in that part on the Ljubljana Moor, which is dotted with numerous prehistoric pile dwellings, some of which are also on the UNESCO World Heritage List. The entire moor is under the special protection by the state authorities and the wooden point was found during a protection survey in the Ljubljanica River. Near the wooden point a flat-bottomed shipwreck from the Roman period was also found [3].

Additional laboratory investigations of the artefact (age and type of wood; unnatural/manual processing of the wood with other tools or fire) were carried out. Radiometric



**Citation:** Guček Puhar, E.; Korat, L.; Erič, M.; Jaklič, A.; Solina F. Microtomographic Analysis of a Palaeolithic Wooden Point from the Ljubljanica River. *Sensors* **2022**, *22*, 2369. https://doi.org/10.3390/ s22062369

Academic Editors: Giovanni Leucci and Marco Zanatta

Received: 12 February 2022 Accepted: 12 March 2022 Published: 18 March 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). analysis (AMS14C) places the point in the Palaeolithic period. Two datings have been made. The first (Miami–USA) showed yew wood (Taxus sp.) to be older than 43,970 years (Beta-252943), and a re-dating in Oxford gave a maximum age of  $38,490 \pm 330$  BP (OxA-19866).



**Figure 1.** The 40,000 years old Palaeolithic wooden point, only one out of eight wooden artefacts of such age known in Europe, was found in 2008, submerged in the river Ljubljanica at Sinja Gorica [2]. As any waterlogged wooden artefact it had to undergo a conservation process to prevent it's complete deterioration once taken out of water [4]. The wooden point before conservation (photo by Arhos d.o.o.).



**Figure 2.** Northeastern area of Vrhnika (Slovenia) with the Nauportus area (a; b) and the location (3) where the Paleolithic wooden point was found in the river Ljubljanica.

Since the discovery, special attention and care have been taken to ensure conditions for the permanent preservation of the point in its natural environment. Underwater archaeologists suggested sinking the point in a special capsule in the deeper sediments of the site in the riverbed of the Ljubljanica. In this way, the point would be optimally protected from the danger of rapid decay. The 3D model and the virtual replica would be presented to the public. However, a different view prevailed. This envisaged the conservation and public presentation of the Palaeolithic remains as a valuable museum exhibit. The lace was kept for five years under artificial conditions (in distilled and regularly changing water). In 2013, the lace was sent to the Römisch–Germanisches Zentralmuseum in Mainz (Germany), which carried out the conservation with melamine resin (C3 H6 N6) in the years 2013–2015. The conservation was completed in 2017. In 2018, the artefact was returned to the Ljubljana City Museum, where it is kept. The wooden point is now one of only eight Palaeolithic wooden artefacts in Europe. However, no other artefact was made in such a beautiful leaf-like shape.

A visual inspection of the point after conservation revealed that it had changed. The point was smaller, lighter in colour and more clearly curved in the area of the planting. A computerised volumetric and geometric deformation analysis of the point cloud of five 3D models (CloudCompare computer software) confirmed this finding. Significant deterioration and deformation was found during the conservation phase. In 2018, we conducted a comparison of five 3D surface models of the Palaeolithic wooden point [5]. The artefact was scanned in 2009–2018 using optical scanners and microtomographic scanners, i.e., before, during and after the consevation process (Figure 3). The comparative analysis of the 3D surface models was performed with the Iterative closest point (ICP) algorithm and the open-source software CloudCompare, which showed that the deformation process had slowed down but was still present. These circumstances prompted us to conduct an additional microtomographic examination of the anatomical structure of the artefact. The ability to compare several 3D models of the same object is very important for broader conservation treatment and the cultural heritage community [6].

3D MODEL	PP-2009	PP-2013	PP-2015	PP-2017	PP-2018
	ZScaner 800	ATOS	ш	ATOS TRIPLE SCAN	Micro XCT400
3D SCANNER			0.		<b>P A 42</b>
	TRIANGULATION 3D SCANNER	STRUCTURED (blue) LIGHT 3D SCANNER			MicroCT SCANNER
ACHIEVABLE ACCURACY	40 µm	8-16 µm			0.6 µm
FORMAT	STL	PLY	PLY	PLY	STL
3D MODEL MADE	Intri d.o.o., Ljubljana	Kompetenzbereich Wiss Zentralmuseums v sodel- Informations- und Mess Applied Sciencem (Main	ZAG Slovenia Slovenian National Building and Civil Engineering Institute		
THE NUMBER OF	86.856	4.941.990	4.228.412	1.557.716	10.255.984
3D MODEL	n	56.9x	48.8 x	18.2 x	118.1 x
	0.8%	48%	41%	15%	100%

**Figure 3.** 3D scanners used to capture the five 3D models of the palaeolithic wooden point in different years.

Based on computer volumetric analysis of the five 3D surface models recorded in different years [4,7], it was found that there were significant changes and deviations after conservation (Tables 1 and 2 and Figure 4). The artefact was visibly bent in the longitudinal direction (Figure 5). Strong bending was also observed in the plant part of the artefact. A slight bending of the artefact point was also confirmed by computer comparison. Its volume decreased considerably. The prevailing assessment was that the resulting surface changes were mainly due to the conservation process (swelling of the wood followed by

rapid drying). A volumetric 3D model, which would reveal the internal structure of the wooden point, should answer the open questions about the changes detected in the analysis of the 3D surface models, such as bending, reduction in volume, change in shape, etc.

**Table 1.** Comparison of volumetric values of 3D models of The Palaeolithic. wooden point (Deformation 2009/2013–2019). \* — Manually measured and visually estimated. \*\* — Digitally measured with *CloudCompare* computer program. \*\*\* — Change (+ –  $\mu$ m) of volumetric value with CloudCompare by measurement before conservation procedure. \*\*\*\* — Change (+ –  $\mu$ m) of volumetric value with CloudCompare by measurement after the first phase of the conservation procedure (irrigation) with melamine resin ( $C_3H_6N_6$ ). PP0—Palaeolithic wooden point. PP3D—Palaeolithic wooden point (3D surface-based model).

	In Situ Ex Situ						
-	3D Models						
3D Dimensions	PP0 2008	PP3D 2009	PP3D 2013	PP3D 2015	PP3D 2017	PP3D 2018	PP3D 2019
and <sup>–</sup> Volume	0 *	1 **	2 **	3 **	4 **	5 **	6 **
	μm	μm	μm	μm	μm	μm	μm
Lenght	160,000	155,606	160,958	152,709	151,768	150,435	149,171
+/- % ***		100	+3.44	-1.86	-2.47	-3.32	-4.10
+/- % ****			100	-5.13	-5.71	-6.54	-7.32
Width	51,000	50,014	52,274	50,594	50,348	48,359	50,705
+/- % ***		100	+4.52	+1.16	+0.67	-3.31	+1.38
+/- % ****			100	-3.21	-3.68	-7.49	-3.00
Thickness	25,000	25,579	28,810	23,856	23,585	22,689	23,793
+/- % ***		100	+12.63	-6.74	-7.79	-11.30	-6.98
+/- % ****			100	-17.20	-18.14	-21.25	-17.41
	$\mu m^3$	μm <sup>3</sup>	$\mu m^3$	$\mu m^3$	$\mu m^3$	$\mu m^3$	μm <sup>3</sup>
Volume –		70,653.6	80,404.1	66,382.8	65,238.9	63,871.9	63,289.4
+/- % ***		100	+13.80	-6.04	-7.66	-9.60	-10.42
+/- % ****			100	-17.44	-18.86	-20.56	-21.29
			C <sub>3</sub> 1	$H_6N_6$			
			CONSE	RVATION			
			BEGIN	END			

A comparative study of 3D surface models of the Palaeolithic wooden point has shown [5] that the 3D models recorded with optical scanners and reconstructed with conventional computer vision algorithms did not contain the necessary information to fully assess the condition of the artefact. The changes to the surface-based 3D models did not answer the question of what state the artefact was in after the conservation process, nor the question of the influence of internal deformations on the surface of the object. These shortcomings can be remedied with a volumetric 3D model that can be reconstructed from 2D tomographic images using computed tomography or computer vision algorithms. **Table 2.** Volumetric changes of the artefact In situ and Ex situ (conservation with melamine resin). \* In our case study there was an increase due to intensive irrigation of the artefact in an aqueous solution (the preparation phase for the conservation process). \*\* Confirmed using a CT scanner.

Protection		In Situ		Ex Situ		
3D Model		PP-2009	PP-2013	PP-2015	PP-2017	PP-2018
			Δ	Δ	Δ	Δ
	Length	_	Enlargement + *	Reduction –	Reduction –	Reduction –
	Width	_	Enlargement + *	Reduction –	Reduction –	Reduction –
	Thickness	_	Enlargement + *	Reduction –	Reduction –	Reduction –
Volumetric	Volume	—	Enlargement + *	Reduction –	Reduction –	Reduction –
parameters +/-	Deformation	_	No	Bending	Bending	
	Degradation	_	No	Crack	Crack	FE0000 Crack/Fracture /Shrinkage/Crumbs /Hole/ **
	Ovality	_	No	Change	Change	Change



**Figure 4.** Index of change in volumetric data after the first scan (2009—index 100). The polar diagram shows the dynamics of volumetric changes (length, width, thickness and volume) between five 3D models of the Palaeolithic wooden point recorded between 2009 and 2018. The red line represents the volumetric state of the point at the first 3D scan (2009—index 100), which reflects the approximation of the state of the artefact under In situ conditions. The diagram clearly shows the changes that occurred at the beginning of the conservation process (2013—black line), when the artefact was subjected to intensive soaking and the addition of melamine resin (conservation). Thickness and volume increased significantly during this phase. After the drying process, the volumetric values (grey, yellow and blue line) decreased, especially the volume and thickness. However, slight dimensional changes in length and width were observed.



**Figure 5.** The deformation changes of the Palaeolithic wood point between the beginning and the end of preservation (volumetric changes, volume reduction and shape change) were calculated with the C2M (ICP) algorithm in the graphical software tool CloudCompare. The left image of the point shows the volumetric changes (red—bending of the upper and plant part; green—shrinking of the middle part) of the surface-based 3D model of 2017 compared to the reference 3D model (2013). The image on the right visualises a change in the shape of the 3D model of 2017 compared to the reference 3D model (2013). The colourimetric scale was created using an algorithm (C2M—CloudCompare) to statistically process the volumetric changes between the 2013 and 2017 models. The red-orange values represent the diffraction of the artefact from +3.6 mm to + 1.2 mm. The blue-green values mark the shrinkage of the artefact between 10.4 mm and 1.9 mm. The data confirms the bending of the handle part and the top of the artefact. However, the shrinkage was more pronounced in the middle part. The cause of this deformation remains unclear. The unclear causes for the changes during the conservation process were the basic motivation for the creation of the anatomical 3D model of the Palaeolithic wooden point. By reconstructing the 3D model from 2D micro-CT images, we wanted to obtain volumetric data on the changes in the anatomical structure of the archaeological object.

A comparison of the amount of information contained in point clouds or triangulation meshes of 3D surface models acquired with optical scanners and tomographs illustrates the qualitative advantage of tomographs (Table 1). This also represents an important difference between the point clouds of the surface-based and volumetric models that can be reconstructed from tomographic images. Optical scanners provide only a small percentage of the information contained in the volumetric model compared to a tomography scanner. This fact is also confirmed by the finding that 3D surface models do not contain comprehensive information about an object. The difference between the two models is also that we can reconstruct both the surface-based and the volumetric 3D model from the point cloud obtained with a tomographic scanner. In contrast, only a surface-based 3D

model can be reconstructed from the point cloud obtained with optical scanners (Table 2, Figures 4 and 5).

Depending on the 3D scanner, the technique for reconstructing the 3D model is also different. In the traditional 3D surface model, we reconstruct the model by combining surface images (texture, volume, colour, etc.) in three-dimensional space. In the volumetric 3D model, we reconstruct a 3D model from the greyscale values (HU or RGB) of 2D tomography images. Such models are more accurate, of higher quality and with less noise compared to conventional computer vision models. They also provide optimal conditions for objective evaluation of the scanned object—volumetric analysis, qualitative analysis of the material, deformation analysis, etc.

#### 1.1. Motivation

The Palaeolithic wooden point, which we have been studying since 2008, was scanned with a microtomographic scanner (Micro XCT 400) in 2018 and 2019. This provided 2D tomographic images that allowed us to look inside the artefact to better understand the reasons for the various deformations that occurred, especially after the conservation process. Analysing and understanding the changes in a waterlogged artefact over a period of 10 years is very valuable knowledge for scientists and conservators. Therefore, we decided to reconstruct surface-based and volumetric 3D models from the available 2D tomography slices. In this way, we were able to investigate the internal cracks, fractures, pores, inclusions and deformations of the wooden point. Since we wanted to have complete control over the tomographic reconstruction of the 3D model from CT sections, we decided to design and implement our own algorithm.

Reconstructing 3D models from CT images is a particular challenge for further research in computer vision, where rendering 3D models with open-source algorithms from images captured by DSLR cameras, structured light scanners, laser and other scanners is widely used for surface and remote sensing. This technique still belongs to the field of computer graphics. A number of segmentation algorithms are adapted to the specific objectives of the users. It is no coincidence that the primary application of noninvasive computed tomography is not the reconstruction of 3D models in the sense of computer vision. The focus is on the interest in noninvasive graphical insights, visualisations and analyses of 2D, 3D or 4D information in a three-dimensional coordinate system for the purpose of visual support in medical diagnostics, measurement technology, analysis and quality control of materials in industry and science.

Most algorithms for processing and reconstructing CT images have not received special attention from computer vision in general or its application in cultural studies due to the emphasis on commercialisation, closed code, specialisation (medicine or industry), radiation risk and intellectual property protection. With this novelty, i.e., the 3D anatomical model and the proposed algorithmic solution, we wanted to present a new challenge to traditional research and applied work in the field of computer vision in heritage sciences. The 3D anatomical model not only enriches the digitised cultural heritage and knowledge about the artefacts, but at the same time expands the possibilities for their analytical treatment, holistic visualisation and augmentation.

Since the background of our research team is in 3D computer vision [8] and we entered the heritage science by helping underwater archaeologists in documenting underwater sites [3,9], this foray into 3D models and tomography seemed promising. We aim to expand the research branch of computer vision with new experiences and insights—using CT, MRI and other depth imaging recording devices—from already-established surface 3D models to the reconstruction of anatomical 3D models and to the use of depth imaging recording devices, which are becoming increasingly affordable on the threshold of the fourth industrial revolution. We aim to complement traditional machine vision, machine and deep learning and artificial intelligence algorithms with noncommercial algorithms and software tools for reconstructing 3D models from CT images. To reconstruct the 3D model from 2D images CT, we selectively used some time-limited open-source and commercial software tools (VolPack, ImageJ, Visualisation Toolkit, InVesalius, 3D Slicer, 3DimViewer and commercial software Avizo Fire, etc.) for 3D rendering and visualisation of tomographic images during the preparatory phase. However, we were not satisfied with the initial results. Our research objective was a comprehensive volumetric reconstruction of the internal (anatomical) structure of a 3D and 4D model of a selected archaeological object. Commercial 3D image processing programmes usually use special segmentation algorithms to process specific 2D image file formats and are designed for use in cultural studies. We often faced limited use of converted files, additional procedures, inaccurate volumetric data and substitution of 3D graphics for 2D graphics in analytical procedures. These circumstances were the reason why we decided to develop a special segmentation algorithm for reconstructing a 3D model from microtomographic images. The algorithm was adapted to the work requirements of archaeological and conservation treatment of smaller archaeological objects.

Archaeologists and conservators occasionally come across CT images of artefacts in their work, but they do not have the appropriate software tools and solutions for noninvasive reconstruction and analysis of 3D models. Since working with expensive hardware and commercial algorithms is not designed for the goals, interests, needs and problems of archaeological or conservation research, the vast amount of data and information contained in 3D anatomical models remains unused and undocumented. At the same time, the use of a customised segmentation algorithm also pointed to new challenges and opportunities in the reconstruction of 3D models in computer vision, where work is still based on more affordable hardware for reconstructions of only surface 3D models. Conventional 3D models of computer vision, compared to 3D models reconstructed from 2D images CT, are of lower quality, highly noisy, volumetrically unreliable and limited both graphically and in terms of data. We have successfully overcome these shortcomings with new, simple and undemanding segmentation algorithms for reconstructing a 3D model from 2D images CT.

### 1.2. Structure of the Article

The rest of the article is structured as follows: Section 2 gives an overview of the use of 3D modelling and computed tomography in archaeology, Section 3 deals with the state of the art in computed tomography in general, Section 4 deals with tomographic reconstruction in the context of considerations in archaeology, Section 5 describes the development of our own software for tomographic reconstruction. We have developed two versions of our software, the first with whole 2D slices CT and the second with segmented 2D slices CT. Section 6 presents the results of tomographic reconstruction with both versions of the software. The software was not only tested on the wooden point, but also on some other archaeological artefacts for which tomographic measurements (CT slices) were available. The internal structure of the wooden point was analysed in detail and various visualisations of this structure are given. The article ends with a discussion and conclusions.

# 2. Background

The London Charter [10] and the Seville Principles [11] recommended 3D models as the standard for archaeological documentation at the beginning of the 21st century. The introduction of new information technologies has fundamentally changed attitudes towards cultural heritage remains. The managers of archaeological collections are increasingly opting for 3D modelling, 3D visualisation or 3D augmentation of archaeological objects. In this way, archaeological objects are protected from the negative effects of ex situ conservation and public display. Archaeologists and conservators receive a more comprehensive professional treatment of the artefacts.

3D models reconstructed with traditional computer vision algorithms have become permanent digital carriers of artefact information that archaeologists can later examine,

evaluate, compare and analyse without damaging or destroying the original objects. However, 3D surface models only contain information about the surface properties of artefacts. This information limitation can be improved by developing and using specialised algorithms to reconstruct volumetric 3D models of archaeological objects from image recordings of deep scanners (CT, MRI, PET etc.) to reveal their internal structure or anatomy as well.

The use of volumetric 3D models represents a qualitative advance in the integration of increasingly affordable computer technologies and software solutions into the regular professional and research work of archaeologists and conservators. The AGORA 3D Heritage Project has also been working towards this (The Belgian Federal Agora3D Project/BELSPO AG /00/164/—Royal Museum of Central Africa—Tervuren—Belgium) [12,13]

Surface-based and volumetric 3D models would greatly enrich archaeological documentation, improve work in archaeological laboratories and enable conservation science to deal comprehensively with cultural heritage remains. However, volumetric 3D models can provide conservators and restorers with additional useful information for planning the use of appropriate methods and techniques to protect critical cultural heritage remains.

# Computed Tomography for Archaeological Documentation

The number of scientific papers dealing with the use of computed tomography (CT) is very extensive in the international research environment. The number of bibliographic units has increased in recent years. This is due to the fact that investment in the development and use of CT technologies is increasing year by year and is becoming one of the most important areas of Industry 4.0 [14]. At the forefront is research into the use of CT in medicine, biology, chemistry, genetics and industry. Interest is also growing in the fields of civil engineering, materials analysis, cultural heritage protection and archaeology. A broader overview of the work of CT shows that the use of CT still raises a number of open questions that also directly affect its place and role in archaeology. These questions indirectly touch on the theoretical basis of our research, such as: optimisation of algorithms [8] for segmentation of CT images [15], development of specific protocols to achieve higher image contrast [16], revival of interest in the use of iterative algorithms in the reconstruction of 3D surface models [17,18], special control, steering and reverse engineering techniques in industry and additive manufacturing [19,20], inconsistent and poor standardisation of CT files [21–23], etc.

A review of published sources shows that the use of nondestructive CT or  $\mu$ CT technology in archaeology has so far focused mainly on the study of very delicate objects from the Palaeolithic and early civilisations. For example, CT scanners have been used to examine mummies [24], Ötzi [25] and The Venus from Willendorf [26], to read scrolls, to study the structure of clay tiles, pottery [27] and Etruscan bronze statues [28], to study textiles, wooden, bone and metal objects, the contents of sarcophagi, the forensic evaluation of art paintings, the technologies used to produce various objects and the mummification process, etc. Archaeologists still predominantly used the technologies, procedures and algorithms of medical computer tomography. It was only after 2015 that there was an awakening of interest in the development of special algorithms and in the use of specially adapted industrial CT readers for mapping earth layers at archaeological sites [29].

After 2019, industrial  $\mu$ CT scanners and specialised commercial software will be increasingly used in archaeological studies (Figure 6). The first steps have already been taken to develop specialised  $\mu$ CT scanners suitable for [30,31] archaeological work. Archaeologists increasingly prefer the use of noninvasive computed tomography methods. With their help, archaeologists can analytically assess the internal structure and properties of smaller archaeological objects.

Current research interests in the noninvasive use, reconstruction, visualisation and processing of 2D  $\mu$ CT images include: analysis of all types of materials [32], analysis of the condition and production method of ancient Egyptian statues for restoration [31], morphological odontometric analysis of dental artefacts [33], visualisation of alloy composition and corrosion exposure of ancient Greek coins [34], evaluation of the suitability of  $\mu$ CT for the study of ceramic manufacturing technology [35], visualisation of the condition of

clothing fabrics [36], analysis of manufacturing techniques and porosity levels of model samples of selected building materials (mortar) in historical objects [37], determination of the condition of the artefact before restoration [38], etc. An interesting example is the development of a system for the simultaneous use of computed tomography and photogrammetry [39] hardware for the visualisation of a 3D model. Important development and research work is being carried out in the computed tomography laboratory of the University of Bologna [30] and in the Centro Conservazione e Restauro "La Venaria Reale" to develop special tomographic devices for archaeological research. (CCR—Turin) [31].



**Figure 6.** Number of articles indexed in Google Scholar and published in MDPI journals during 2015–2021/11 that represent the use of computed tomography in the field of cultural heritage science and are directly or indirectly related to the topic of 3D rendering from 2D tomography images or 3D slices. No article was dedicated to the problem of direct reconstruction of 3D models from 2D tomography images. The 3D models were reconstructed using commercial industrial tomography software (VGStudio MAX [31,34,35,37], Amira Avizo 9.0 [32,33], Simpleware—Synopsys, Inc, Dragonfly Pro—Carl Zeiss) and in most cases analysed from 2D slices in different layers. In no case did we record the use of any of the open source programmes to represent a 3D anatomical model. Open source programmes do not currently provide data for advanced statistical and geometric analyses.

The current importance and applicability of computed microtomography in analytical archeology is confirmed by the microtomographic treatment of the Venus from Willendorf by the Department of Evolutionary Anthropology and the Core Facility for Micro-Computed Tomography, University of Vienna, which researched and microtomographically processed the approx. 30,000 year old figurine made out of oolitic limestone using commercial computer software Amira. The research offered important new insights into the origins and methods of making this valuable archaeological artifact [26].

# 3. State of the Art in Computed Tomography

The original dilemma of algorithms for processing 2D images for CT at "low-level" was solved decades ago in the framework of Marr's "high-level" paradigm [40], although computed tomography algorithms for processing X-ray signals and 2D images still have to deal with the problem of optimising edges and segmenting 2D images. The problem of robustness is obvious. Algorithms should primarily be based on the paradigm of active detection [41], i.e., they must be targeted at specific goals of detecting certain features, such as edge detectors, region growers, 3D recovery methods, etc.

The use of computed tomography and tomographic sensors is increasingly expanding the field of robotic surgery [42–44], artificial intelligence [45], computer graphics (e.g., 3D Geometry Generator Algorithm [46]) and computer vision (e.g., PT2PC model [47,48]).

In computed tomography we distinguish two types of algorithms, depending on the type of reconstruction. Figure 7 shows the process diagram of algorithms for spatial and surface imaging:

- 1. algorithms for the reconstruction of 2D tomographic images from tomographic projections and
- algorithms for the reconstruction of volumetric 3D models from 2D tomographic images. For the reconstruction of 3D models from CT images, the terms "volume rendering" and "surface rendering" are often used synonymously in computer graphics [49,50].

However, the use of both terms requires some caution as they are often used uncritically and indiscriminately in professional articles, even in cases that do not involve reconstruction of 3D surface or volumetric models from CT, MRI or PET images and are based on attenuations or RGB matrices of grey values. The term "volumetric imaging" has not yet gained acceptance in computer vision methods and techniques for reconstructing 3D models. However, it is accepted in computer graphics methods and techniques for 3D visualisation of tomographic images [51].

While in the first case, the algorithms are adapted and specialised for processing image signals based on attenuation values or HU numbers, in the second case various specialised algorithms are used to represent the 3D surface model. These include algorithms from computer vision and computer graphics [52,53], such as the Marching Cubes algorithm, the dividing cubes algorithm, algorithms for visuomotor and haptic rendering [54].

The reconstruction of CT images in three-dimensional space is of particular importance in modern medicine and industry. Especially in robotic surgery, prosthetics and orthopaedics as well as in the analysis of material quality. This is not only about real-time visualisation and improving accuracy in the detection and treatment of diseases, such as in robotic surgery, but also about achieving higher quality products in industry. In both medicine and industry, methods for reconstructing and visualising 2D tomography images are on the rise. Algorithms for target rendering are also being adapted and developed accordingly. For example, Microsoft Visual Studio [49] has developed four algorithms with the Visualisation Toolkit (VTK) to help with robotic surgery: Marching Cube, Contour Filter, Composite Volume Rendering and Texture Mapping Hardware. Marching Cube and Contour Filter are algorithms for surface mapping, while Composite Volume Rendering and Texture Mapping Hardware are for spatial mapping. After 2000, similar development trends can be observed among other manufacturers of commercial hardware and software for computed tomography. The programming languages used to implement the algorithms are C++, Java and Python. However, it is undeniable that algorithms for surface or spatial imaging are adapted to specific needs and that the question of their robustness is still relevant.

In recent years, there has been increasing interest in the use of machine learning and deep learning algorithms, convolutional neural networks (CNN) or other artificial intelligence algorithms in industrial computed tomography. Occasionally, adapted or modified algorithms are used that have already been used in the reconstruction of 2D



tomography images, such as Fourier volume rendering, Monte Carlo volume rendering [55], the additive re-projection technique, depth-shading algorithms, etc. [52].

**Figure 7.** Computed tomography—presentation of the process of reconstruction of 2D CT images and 3D models.

Algorithms for imaging or reconstructing 3D volumetric models can be divided into four groups: Ray Casting Algorithms, Splatting Algorithms, Cell Projection Algorithms and Multi-Pass Resampling Algorithms. However, in the practice of 3D visualisation and modelling from tomographic images, three techniques for volumetric image projections in computed tomography stand out: maximum intensity projection (MIP), minimum intensity projection (MiniIP) and three-dimensional volume rendering (3DVR) [56] or direct volume rendering (DVR) [51].

An optimised 2D tomographic image is a key factor for the quality of all further procedures from visualisation to reconstruction of 3D models and addition. Therefore, most research efforts in computed tomography have been and still are focused on the improvement of algorithms for 2D image reconstruction and not on the reconstruction of 3D models. In computed tomography, computer graphics algorithms have become the most popular. Chris Shaw [57] mentions that so far (2014) only a modest 1.8% of research papers have been published that deal with CT and are also intended for 3D imaging.

A transparent contribution to the understanding and open questions of computerised tomography and the algorithms used for reconstruction was presented by G.T. Herman in his most cited work in the field of computerised tomography, his book *Fundamentals of computerised tomography: image reconstruction from projections* [58]. Herman describes how CT image data are obtained and used in science and medicine. The focus is on X-ray data, but also on the importance of CT in other fields, such as electron microscopy, nuclear medicine, ultrasound, magnetic resonance imaging, nondestructive testing and evaluation of materials, etc. A comparative evaluation of the reconstruction algorithms are also covered, including the filtered back projection, the extended Fourier theorem for reconstruction and back reconstruction, the linogram method for image reconstruction,

algebraic reconstruction techniques, quadratic optimisation, etc. The book also draws attention to the open questions and problems of CT image reconstruction.

A second review of the features, capabilities and shortcomings of segmentation and reconstruction algorithms in computed tomography can be found in *3D Segmentation Algorithms for Computerised Tomographic Imaging: a Systematic Literature Review* [59]. The article includes peer-reviewed articles published in four scientific databases (Science Direct, IEEEXplore, ACMin PubMed) from 2006–2018. For the authors (Carvalho, Sobieranski and VonWangenheim), the key to the reconstruction process in computed tomography is the segmentation algorithm. A total of 182 papers were analysed, divided according to the methods and segmentation techniques used, namely thresholding methods, graph theory, level set methods, Markov Random Fields, active contours, flexible point distribution, wave densities, region growing, primitive shapes, use of filters and histograms, intelligent swarms, up to the use of convolutional neural networks, deep learning and machine learning. An interesting example is a method that combines segmentation and visualisation of multiple anatomical structures [60]. There is a very wide range of algorithms.

In tomographic image reconstruction in medicine, additive manufacturing, materials analysis and industrial control, the filtered backprojection (FBP) algorithm has been standardised for some time. In recent years, somewhat forgotten iterative reconstruction algorithms have resurfaced in industrial tomography [61–65]. Their use has become interesting with the increasing computing power of computers. Comparisons and research show certain advantages of the reconstruction algorithms over the FBP algorithm. New iterative reconstruction algorithms such as AIDR [62], ASIR and ASIR-V [66], IRIS [63,64] SAFIRE [65], ADMIRE [67], etc., show certain advantages over the FBP algorithm. Namely, iterative reconstruction significantly improves image quality due to cyclic processing. In medical tomography, iterative algorithms also mitigate the negative consequences (noise, artefacts, quality, sharpness) resulting from the requirement of selective use of the radiation dose index [62]. New iterative algorithms are already built into the latest generations of CT readers (e.g., Siemens, Toshiba, etc.) and are in most cases a trade secret.

The development so far shows a multitude of specialised reconstruction algorithms. The number and variety of algorithms is due to the fact that each selected algorithm and segmentation method or technique is adapted to specific research questions of 2D tomography image reconstruction in medical or industrial diagnostic imaging or analytical practise.

Computed tomography algorithm developers are still mainly focused on optimising algorithms for segmentation, recovery and 3D reconstruction of CT images [8,59] and on creating and using technological protocols or computational methods to achieve higher contrast and quality of CT layers in 3D visualisation [16]. However, research in iterative reconstruction (BIR, SiR-V, etc.) is still ongoing [17,18]. The transition from slice imaging to volumetric imaging and direct 3D reconstruction with spiral computed tomography (SCT) [68] has led to significant advances in medicine and industry in recent years. SCT also opens up new possibilities in the realisation of virtual archaeology projects.

#### 4. Tomographic Reconstruction in Archaeology

Traditional radiology after 1889 and computed tomography after 1975 have been present in archaeology since its beginnings. The use of algorithms for the reconstruction of 2D images (for example: Inverse Fourier Transform or the filtered back projection commonly used today) is considered a noninvasive technique for the anatomical study of delicate, unstable and valuable artefacts [24] such as mummies, stone, Palaeolithic bone or wooden remains [4,5,7,69], papyrus scrolls, metal tools or weapons [28], jewellery, pottery, wall paintings, Ötzi—Italy [25,70], the Venus from Willendorf [26] etc.

Only rarely do we find examples of the reconstruction of archaeological 3D models from tomographic images. This is partly due to the development of a relatively autonomous field of computed tomography, which is primarily developed as an imaging technique in medical diagnostics and as a measurement and control technique in industry. The algorithms for reconstructing the X-ray signal into a 2D image and 3D visualisation from CT images are also geared towards these goals. Isolated attempts at 3D modelling from CT images are also the result of the high financial cost of using CT scanners and the still insufficiently exposed need to use 3D anatomical models in additive manufacturing.

The reconstruction of 3D models of archaeological artefacts has so far been mainly limited to 3D surface-based modelling in archaeology. 3D surface-based models have in most cases been created using affordable photogrammetry, laser scanners and structured light scanners. Various computer vision algorithms have been used for reconstructing 3D surface models, such as: intuitive algorithms for computing similarity or distance [71], SIFT— Scale Invariant Feature Transform [72,73], SURF—Speeded Up Robust Features) [74,75], ICP [76], SfM [76,77], SfS [78], SfL [79], segmentation algorithms [80], stereoreconstruction algorithms [81,82], self-learning algorithms [83], etc. In the last year, the use of artificial intelligence algorithms (deep learning, convolutional neural networks, etc.) has come to the fore. This is also the reason why the use of information technology in archaeology has focused on virtual archaeology [10,11], the additive production of copies of artefacts using surface-based 3D models and the digitisation of basic archaeological documentation. The processing, evaluation and comparison of the digitised information in archaeological document collections was left to future generations of archaeologists.

The reconstruction of anatomical 3D models from tomographic images is not yet the subject of interest of heritage sciences for a comprehensive documentation of archaeological objects. The use of computed tomography algorithms in archaeology still primarily adapts to the possibilities of arbitrary use of medical and industrial hardware. It is only since 2015 that we have seen greater interest from archaeologists and conservators in the use of computed tomography in the processes of conservation, restoration and visualisation of archaeological objects [84–88].

According to studies published in 2020 and 2021, archaeologists and conservators continue to use commercially tested material and composite analysis software tools to reconstruct or visualise 3D archaeological models, such as VGStudio MAX, Amira Avizo 9.0, Simpleware—Synopsys, Inc. and Dragonfly Pro—Carl Zeiss. We also recommend the possibility of online 3D visualisation of 2D images CT, with [89] interactive 2D and 3D graphics (WebGL), in any compatible web browser and without the use of plug-ins.

This is also the reason why, given the still financially inaccessible hardware, no specific computed tomography hardware or algorithms have been developed for the needs of heritage conservation, suitable for documentation and research science. Such initiatives have been presented in the past. For example, the Belgian project AGORA3D was launched in 2008. In recent years there have been efforts in this direction, for example by researchers at the University of Bologna and the Conservazione e Restauro "La Venaria Reale"—CCR of the University of Turin. An important innovation in the use of computed tomography for the conservation of valuable remains of cultural heritage are interdisciplinary projects of some French (e.g., the Introspect project) [84,88], British (RTISAD project), American (e.g., EDUCE project; Mummy project) [87], Canadian, Israeli, Austrian [86] and German [85] university research centres. In collaboration with specialised laboratories of state museums and some private companies, they use computed tomography (CT algorithms) in the planning, conservation and restoration of museum and archaeological exhibits.

# 5. Development of a New Tomographic Reconstruction Tool for Archaeology

The basic research objective was to select, improve and evaluate an algorithm for highquality reconstruction of 3D surface and 3D volume models from tomographic images for the needs of preservation, conservation and evaluation of archaeological heritage remains. With the new approach in computer vision, mainly using algorithms for 3D surface-based modelling in the processing of 2D image records, we wanted to highlight the suitability and applicability as well as the analytical and documentary importance of 3D volumetric models in computer vision and heritage science. At the same time, we wanted to answer some open questions about the observed changes in 3D surface-based models of Palaeolithic wooden point from its discovery to the completion of conservation [2,4,5,7]. As open-source algorithms are used for observing 3D surface models obtained from image acquisitions from DSLR cameras, structured light scanners, lasers and other surface and remote sensing devices, reconstruction of 2D images from CT, MRI and other depth sensors also requires appropriate observational techniques. Most algorithms for processing and reconstructing CT images have not received special attention from computer vision in heritage science due to a focus on commercialisation, closed source, specialisation (medical or industrial), radiation risks and intellectual property protection. With new methods for reconstructing surface and volumetric 3D models from CT images, we aim to challenge the traditional research and application field of computer vision in heritage science.

With the new experiences, we aim to expand the research interest of computer vision (using CT, MRI and other depth-imaging scanners) from the already-established traditional 3D surface models to the reconstruction of surface models and 3D volumetric models from tomographic images. We anticipate that the price of 3D depth detection scanners will become more affordable for cultural scientists on the threshold of the fourth industrial revolution.

A particular research and development goal was to develop a simple algorithm for reconstructing volumetric 3D models from CT images of archaeological objects. The algorithm should not require highly specialised computer and mathematical knowledge from the user. We were guided by the idea that the algorithm should be simple, robust and adaptable to the specific needs of archaeologists and conservators. After all, they occasionally come across CT images of artefacts in their work, but they do not have the appropriate algorithms, customised software tools and solutions for reconstructing and analysing 3D models. Therefore, they are usually forced to use commercial software tools developed and adapted for other purposes, namely diagnostic work in medicine and industry. Since working with them is not tailored to the goals, interests, needs and problems of archaeological or conservation research, the huge amount of data and information contained in surface-based and volumetric 3D models remains unused. This leads to an impoverishment of the digital documentation of cultural objects. Developing a simple computer algorithm suitable for working with cultural heritage remains was our ultimate motivating goal, which would provide an incentive for further research in computer vision and cultural heritage science.

#### 5.1. Design Considerations

After studying the literature and initially testing various computed tomography, computer graphics and computer vision algorithms for reconstructing 3D models from 2D images CT, we were faced with the need to develop a new iterative algorithm. The tested solutions did not meet our expectations to reconstruct a surface-based and volumetric 3D model that would answer the open questions about the identified changes in the five surface-based 3D models of the Palaeolithic wooden point accurately, volumetrically precise and graphically clear enough. The test with commercial software also showed that the built-in algorithms reconstruct the 3D model, but only the outer surface edge points of the cloud. Volumetrically, it reliably and accurately reconstructs only the 3D surface model of the scanned object.

The workflow of the new algorithm is based on a preliminary analysis of the properties of the 2D CT image slices and the projection of the output file of the 3D model (an OBJ file) based on the following parameters of the reconstruction of the surface-based and volumetric 3D model:

- 2D image CT in the format TIFF (JPEG, GIF, PNG, BMP, RAW, etc.) consisting of a grey HU or RGB matrix in which each point (pixel) of the matrix has a grey x/y value;
- the thickness of the X-ray beam corresponding to the z-coordinate value of each voxel of the 3D layer in Cartesian three-dimensional space;
- an edge-detection technique that segments selected greyscale RGB values of each boundary point (pixel) of a 2D image.

The algorithm must be adapted to the purpose of the archaeologist or conservator to reconstruct the 3D surface and 3D volume model of the artefact under investigation. In the selected test case this will be: detection of dislocations, inclusions, pores, cracks, openings, damage, deformations and fractures in the anatomical structure of the object.

The design of the algorithm should focus on the specific research objective of reconstructing the 3D surface and 3D volume model in the initial phase of development. Based on the available data contained in the greyscale values of the images from CT, we found that the reconstruction of the 3D spatial and 3D surface model can, on the one hand, use the full set of attenuation or RGB values and, on the other hand, develop an algorithm that can detect features in the anatomical structure of the object that are aligned with a specific archaeological or conservation goal.

Methodologically, two iterative algorithms were developed that differ in terms of the degree of robustness, the research purpose and the temporal dynamics of the implementation:

- the algorithm with the working code name *dAR3D* converts 2D images from CT directly and without additional segmentation into 3D slices CT (3D scalar field of voxels), registers them in a three-dimensional coordinate system as a collection file and then reconstructs them into a volumetric and surface-based 3D model, and
- the algorithm with the working code name *sAR3D*, which first limits the number and values of features in the 2D image CT by segmentation. It then converts the segmented 2D images CT with a *z*-coordinate value into a 3D scalar field of voxels (3D CT slices), registers them in a three-dimensional volume coordinate system as a collection file, and then reconstructs them into a 3D volume and 3D surface model using the aggregation and alignment method. The segmentation algorithm is intended for specific analysis or research goals in the archaeological, conservation or restoration treatment of an archaeological object. It is faster than the *dAR3D* and can be implemented on personal computers without memory limitations. It can also be used in other fields or in cases where we decide to segment features in a 2D CT (MRI etc.) image.

The algorithms are limited to the process of reconstructing the 3D surface model and the 3D volume model from 2D images CT (Figure 7).

The direct algorithm *dR3D* is slower (due to the large amount of information contained and consequently the size of the output file). Reconstructing a 3D model takes about 7 times longer than reconstructing a 3D model from segmented CT images. Due to the large amount of information, additional computer storage capacity or cloud processing is required. This algorithm is useful:

- in the initial nonselective and noninvasive examination of the anatomical structure
  of the artefact to obtain initial information about the artefact and for the subsequent
  selection of target features for archaeological, analytical or conservation processing;
- for a comprehensive archaeological 3D documentation, and;
- to produce a 3D additive as a perfect replica of the original.

The segmentation algorithm *sAR3D*, which was the subject of our special development refinement due to its robustness, simplicity and efficiency, is adapted to the research goals of the direct user. It is intended for the working purposes of archaeological, conservation or restoration treatment. It is faster in time and can be realised on noncommercial personal computers without memory limitations. It is user-friendly, robust and simple. No special knowledge (radiology, mathematics, physics, computer vision, computer graphics, computer tomography, etc.) is required to achieve the user's research goals of reconstructing a 3D model from CT images.

The workflow of the two algorithms is shown in Figure 8.



**Figure 8.** Workflow of the direct *dR3D* and the segmentation algorithm *sAR3D* for reconstruction of 3D models from CT images.

# 5.2. Implementation

Code the algorithms for direct and segmented reconstruction of 3D models from 2D  $\mu$ CT (or any 2D images obtained with CT,  $\mu$ CT, nano- CT, MRI, MMG, ultrasound or other depth sensor readers) in four steps. The individual steps and specifics of the code record structure are shown in the tabular overview (Table 3), which lists the basic code records.

In the mathematical and logical design of the code of both algorithms, we used a numerical analysis software package, MATLAB. The chosen fourth-generation programming language is sufficiently effective for performing mathematical operations. However, the segmentation process is only one form of mathematical algorithm. In addition, the programming language can quickly work with tables to quickly convert segmentation data into a 3D coordinate table.

The code of both algorithms is the same in the first step (selecting and registering 2D images CT for 3D reconstruction), in the phase of converting 2D images into 3D layers and in the phase of writing a 3D model in OBJ format. The codes of the two algorithms differ only in the fourth step, where we use the outer and inner edge detection technique to segment individual 3D layers.

The code uses the auxiliary algorithm of the NATSORTFILES function (built into MATLAB SORT), which classifies an array of cells with filenames or paths. It takes into account all numeric values in the array (instead of the default sort, it uses "natural sorts"). It is a "natural sort type" or an "alphanumeric type". NATSORTFILES (74) does not mean sorting by natural order, but separates file names and their extensions. This has the effect that shorter file and directory names are always placed before longer names, thus ensuring the correct order of names in the directory. For the same reason, file paths are divided into each path delimiter and each directory level is sorted separately.

A special feature of the code of both algorithms is the input of "z-values" or "zcoordinates". This procedure is crucial for the correct volumetric arrangement of the 3D model. If we want to reconstruct a surface and volumetric 3D model from CT images that allows accurate volumetric analysis and comparison, it is not enough to simply fold 2D images into a virtual 3D model. To achieve a volumetric match with the original, each 3D layer must be inserted into a Cartesian coordinate model of the space with precise voxel dimension data (the "z-coordinate value" in the 3D model space). Only in this way can we obtain a volumetrically accurate 3D anatomical model that is volumetrically identical to the original. If we simply glue the images of CT together, we will obtain the correct values for the width (x) and thickness (y) of the object, but the length (z-coordinate) will be completely distorted, unrealistic and useless for volumetric analysis and final reconstruction.

**Table 3.** Code characteristics of the segmentation algorithm *sAR3D* for the reconstruction of 3D models from CT images.

sAR3D Code Characteristics					
Step	Code comment	Slide Master			
1	Preparation of the algorithm	<pre>imagefiles = dir('*.tif'); n = natsortfiles((imagefiles.name));</pre>			
	Defining, selecting and sorting image file names; scale;	<pre>nfiles = length(n); scale = 0.053;</pre>			
	file	<pre>fid = fopen('my.obj','wt');</pre>			
2	Loop (go through images by file name)	<pre>for ii =1:nfiles end</pre>			
3	Opens and reads each image file	<pre>currentimage = imread(currentfilename);</pre>			
4	Obtaining and determining the $x - y$ coordinate of points	Segment images and determine the coordinates of points from them			
5	Add the third $(z)$ coordinate to the points	<pre>z = repmat((ii * scale), [size(row,1) 1]);  points = [xy, z];</pre>			
6	Write to specified end file (3D coordinate table)	<pre>result = cat(2,vert,string(points)); fprintf(fid, '</pre>			
7	Close the final 3D model file	Segment 3D model file			

The fundamental difference between the two algorithms lies in the introduction of the segmentation procedure, which in our case is based on the technique of detecting edges on the surface and in the anatomical structure of the studied object. The segmentation procedure removes insignificant grey values of edges from the image CT and determines the edge values that will be the subject of the user's examination. The segmentation procedure allows us to highlight the selected target features of the anatomical structure of the original more clearly in the reconstruction of the 3D model than with the dAR3D algorithm. In this way, we simplify and shorten the reconstruction and extract the features highlighted in the 3D anatomical model while preserving all volumetric values and information.

#### 5.3. Segmentation

In our case, segmentation is considered as a form of mathematical algorithm. This was also the reason why we chose a numerical analysis software package (MatLab) to write the code. This programming language is suitable for working with tables and can effectively convert segmented data into a table with 3D coordinates. The code written in this programming language is generally robust, easy to understand and use, optimal for processing image files (e.g., TIF, JPEG, PNG, etc.) and does not require additional and intermediate procedures.

The right choice of edge-detection techniques is crucial for efficient segmentation. The edge of a CT image can be defined as an abrupt change in intensity. The edge is nothing more than the boundary of a particular region. In a CT image, the edge can be defined as a break in brightness or a change in the grey values of the HU or RGB matrix. When

the light intensity or the grey value changes, this is called an edge. Edge perception is important for describing the shape and anatomical structure of an object. Edge detection identifies boundaries in a 2D image CT. The features can be common or separate. The main goal of edge detection in the segmentation algorithm is to obtain the volume and surface features of the object. Before selecting the segmentation code, we investigated and evaluated different edge detection techniques (Figure 9) and selected the most suitable one for reconstructing a 3D model of an archaeological object.

EDGE DETECTION TECHNIQUES						
CT IMAGE (original)	(Selection of segmentation function) RGB GRAY IMAGE (RGB=130)	ROBERT				
PREWITT	SOBEL	LOG				
ZEROCROSS	CANNY	APROXCANNY				

**Figure 9.** Comparison of different edge detectors for segmentation of 2D slices CT. The figure shows the test results of the different edge-detection methods. The analysis and comparison of the test results was the basis for selecting the most suitable operator for performing the segmentation process in the phase of preparing the 2D micro-CT images for placement in 3D space and reconstruction of the 3D model. The Roberts Edge Operator was selected as the most suitable operator for performing the segmentation process. The advantage of the Roberts segmentation function is that it maintains the most small details without altering small cracks or indentations (these differences are more noticeable at higher magnifications).

We have investigated, compared and tested different techniques (Roberts, Prewitt, Sobel, Log, Zerocross, Canny and Approximate Canny) for edge detection in the image CT of the Palaeolithic wooden point and some other artefacts (Figure 9). We chose Roberts edge-detection technique, which we found most suitable for processing CT images of archaeological artefacts. This technique is simple, robust and does not smooth edges compared to other edge detectors. It is also capable of detecting smaller openings. It uses a  $2 \times 2$  convolution mask. The Roberts operator [90–92] is widely used for image processing in computer vision. It is one of the oldest operators and was introduced by Lawrence Roberts in 1963. The operator is also known as Roberts' cross operator. The edge image can be calculated quickly and easily. Although it is the oldest and simplest, it has proven to be the most reliable in reconstructing a 3D model of an archaeological object compared to other techniques.

The process of reconstructing the 3D surface model and the 3D volume model from CT sections was adapted to the specific research objectives of the *archaeological or conservation treatment* of the artefact using the segmentation *sAR3D*. Segmentation in our case is a technique for determining, selecting and limiting the grey RGB values in CT images. This also allows us to reconstruct the 3D surface and 3D volume model. It is a rarely used 3D

modelling technique, even when it comes to the commercial reconstruction of a 3D model from CT images. However, CT image segmentation is different from the prevailing 3D modelling techniques in computer vision, where segmentation is based on other paradigms, such as: light sources, contrast, colour shading, generalisation of geometric shapes, etc.

In our particular case, segmentation was used to determine and mark the edges of the non-oody x–y–z coordinate points (voxels) in each 3D layer. This marked the features in the anatomical structure of the 3D layer (the outer edge of the artefact, the edges of the anatomical—in our case—nonwoody features of the object). We selected those features in the 3D layer which, in accordance with the aim of the study, we believed could successfully answer the question about the nonwoody features of the artefact and the factors that might influence the observed surface deformation of the object. In our case, these were openings, pores, fractures, inclusions, cracks, etc. By determining the outer edge handles, we have set the stage for the reconstruction of the 3D surface model, while the 3D anatomical model will volumetrically mark and represent selected features that were the subject of research interest and observation.

# 5.4. Visualisation of Results

The visualisation of the 3D model, which was reconstructed from 2D images CT, is done with the graphical open-source software tools CloudCompare and MeshLab, which enable high-quality and accurate visualisation of 3D models, their volumetric analysis and comparison.

The visualisation process is simple. It is done in six steps using the tools, methods, techniques and algorithms built into the software. The following is a summary of the main procedures for visualising 3D models with CloudCompare (Figure 10):

- **Step 1:** the point cloud of the 3D model in the selected format (OBJ, PLY, STL, etc.) is entered using the "Open" function in the CC software;
- **Step 2:** The tool CROSS SECTION removes unwanted or disturbing sections in the point cloud and then exports the selected section as a new point cloud to CC;
- **Step 3:** The calculation of NORMAL with the surface approximation method follows (it is possible to select the methods depending on the goal of the visualisation: planar; triangulation; square);
- **Step 4:** The calculated normals are aligned to the normals using an orientation algorithm or a numerical Fast Marching method (or using the Minimum Spanning Tree algorithm);
- **Step 5:** Poisson Surface Reconstruction is used to convert a point cloud into a 3D model with triangulation mesh.

Depending on the exposed research objectives and the planned treatment of the 3D model and the objectives of its visualisation, follows.

**Step 6:** For further processing and clearer visualisation, various filters, stereogram analyses and other tools are available for multilevel volumetric, statistical and analytical treatment of the point cloud (e.g., cross sections, 3D depth views, comparisons, smoothing, volumetric and statistical tools, segmentation, scalar fields, etc.). In our case, we opted for anatomical or volumetric 3D visualisation and volumetric detection (*x*, *y*, *z*) of critical points in the anatomical structure of the Palaeolithic point.

By processing the triangulation mesh of the points of the visualised model, we can successfully perform an accurate volumetric and deformation analysis and comparison (deformation monitoring) of the selected 3D models.



**Figure 10.** 3D model visualization process in the CloudCompare software tool. Read the explanations of the six steps in the text of the article.

# 6. Results

Algorithms for reconstructing 3D models from CT images were tested on a 40,000year-old Palaeolithic wooden hunting weapon found in the Ljubljanica River (Slovenia) in 2008 [2]. To obtain an assessment of the quality of the represented 3D volume and surface model, the algorithms were tested on four other archaeological objects (bone flute from Divje babe I [69] and three ceramic rattles from the Bronze Age [93]). The robustness of the algorithms was also tested on two composite materials (concrete and fabric). The quality of the reconstructed 3D models was compared with models visualised with the commercial software tool Avizo Fire (FEI).

Volume and deformation analyses with the software CloudCompare were carried out for 3D volumetric models of the Palaeolithic wooden point.

# 6.1. Input-Output Data for the Reconstruction of 3D Models from CT Images

Algorithms used for reconstruction of 3D models from CT Images:

- *dAR3D*—direct algorithm for 3D model reconstruction;
- *sAR3D*—segmentation algorithm for 3D model reconstruction.

Input data (Table 4): reconstructed two-dimensional microtomographic images of an archaeological object in the format TIFF, created on the basis of a matrix of attenuation values (HU -number) and filtered back-projections with a MicroXCT 400 scanner.

Output data (Table 5): 3D model in format OBJ. Displayed and processed with the software tools MeshLab and CloudCompare.

# 6.2. Hardware and Software

The algorithms were systematically developed, tested and evaluated on a commercially available portable computer (a laptop) with the following configuration:

- PROCESSOR: INTEL (R) Core (TM) i7-8850 U @ 1.80 GHz;
- RAM: 8 GB;
- GPU: NVIDIA GeForce RTX 1050;
- OS: Windows 10 (64-bit).

MatLab (ver. R2018a—a custom multiparadigm programming language and numerical computing environment) was used to develop the algorithms.

Open source applications for graphical, statistical and volumetric processing Mesh-Lab and CloudCompare were used to process image data, compare and reconstruct the 3D model.

Artefact	Number	Format	Image Size	Slice Thickness
Input Data	µCT Images			
Palaeolithic wooden point	2452 (year 2018) 2650 (year 2019)	TIFF	$2699 \times 2731$ $1012 \times 1024$	44.2 μm
Bone flute from Divje babe I	2649	TIFF	732 × 837	31.9 µm
Ceramic rattles	R1—1717 R2—1014 R3—1013	TIFF	$\begin{array}{c} 1012 \times 1024 \\ 1012 \times 1024 \\ 1012 \times 1024 \end{array}$	51.9 μm 44.5 μm 62.7 μm
Different composites	B—1014 T—1014 (300)	TIFF	1012 × 1024	44.5 μm

Table 4. Input data for the reconstruction of a 3D model from CT images.

Table 5. Output data of the reconstructed 3D model from CT images.

Artefact		3D Model—File Size	
	dAR3D	Format	sAR3D
Palaeolithic wooden point	8.18 GB (year 2018) 7.7 GB (year 2019)	OBJ	196 MB (year 2018) 193 MB (year 2019)
Bone flute from Divje babe I	4.68 GB	ОВЈ	166 MB
Ceramic rattles	R1—5.12 GB R2—3.26 GB R3—2.49 GB	OBJ	R1—132.0 MB R2—89.2 MB R3—68.1 MB
Different composites	B—14.9 GB	OBJ	B—280 MB
tomp offee	T—1.8 GB	OBJ	T—69 MB

# 6.3. Reconstruction, Comparison and Analysis of 3D Models of the Palaeolithic Wooden Point

The reconstruction of the 3D models of the Palaeolithic wooden point was carried out using both developed algorithms (Figure 11). Based on the obtained results and volumetric data of the point cloud in 3D Cartesian space, we performed an analysis of the anatomical structure and properties of the 2019 3D volumetric model and compared the obtained data and information (deformation monitoring) with the 2018 model. The visualisation and comparison were performed with the graphical software tool CloudCompare.

#### 6.3.1. Reconstruction with Algorithm dAR3D

Using the direct algorithm (*dAR3D*), we reconstructed the 3D surface-based model and the 3D volumetric model directly from the entire set of 2D images CT of the Palaeolithic wooden point (Figure 12) as captured in the HU matrix by the CT reader and recognised by the RGB matrix.

Compared to the original and the model reconstructed with the commercial tool Avizo Fire (FEI), the surface-based 3D model is extremely accurate and even surpasses the photo of the original in certain parameters (contrast, properties of the surface texture). The difference in contrast and accuracy between the reconstructed 3D model and the original photo is due to the limitations of the DSLR camera such as colour gradations, the influence and direction of light on the artefact and other circumstances of the photo. The difference in contrast between the reconstructed dAR3D model and the model displayed with the commercial Avizo Fire (FEI) tool is due to the use of different 2D image resolutions in the TIFF format. In our case, the image resolution used to reconstruct the 3D model was significantly higher (2690  $\times$  2731) than the image resolution (1012  $\times$  1024) used by the commercial tool to display the 3D model. This is also the reason why our model is more accurate and the point cloud also contains a much larger set of stored information.



**Figure 11.** Reconstruction results: (A) direct algorithm *dAR3D*, (B) segmentation algorithm *sAR3D*.

6.3.2. Reconstruction with the Segmentation Algorithm sAR3D

For the implementation of the segmentation algorithm *sAR3D*, we first marked only the grey RGB values (–278 HU; 110–130 RGB) in the 2D images CT for the reconstruction of the 3D surface model and the 3D volume model, as well as regions that indicated nonwood features or values in the images. With this intervention, we limited the segmentation to selected features in the anatomical structure of the point (Figure 13), such as dislocations, inclusions, pores, cracks, openings, damage, deformation and fractures. According to archaeologists, conservators and restorers, these deformations are important for planning procedures to protect artefacts and for further 4D analyses (deformation monitoring) of 3D volume models of the point (2018 and 2019).

The properties of the surface model reconstructed with the algorithm *sAR3D* are identical to the results described and summarised in the model presented with *dAR3D*. Since the segmentation algorithm limited the set of point clouds and the target information in CT images of the point's anatomical structure to the RGB values of the nonlignified parts of the point (pores, openings, cracks, fractures, etc.), only the segmented values in the anatomical structure were highlighted in the 3D volumetric model (Figure 14). The model accurately and clearly represents the changes, features, characteristics and deformations of the artefact.



**Figure 12.** Example of a 3D model reconstruction with the algorithm *dAR3D*. The model detects anatomical changes (cracks, fractures, etc.), but the anatomical structure is also filled with woody parts. The reconstruction of the 3D model was performed considering all RGB values (0–255) of the grey matrix of the 2D images. Deformations of the internal structure of the model are only visible after individual sections. The model does not provide a detailed 2.5D insight into the artefact despite the large amount of information and data.



**Figure 13.** In the 2D CT image the edges of larger openings, fractures, pores, inclusions, etc., are detected first.



**Figure 14.** Palaeolithic wooden point: **(A)** 3D anatomical (volume) model, **(B)** 3D surface model, **(C)** 3D anatomical model with marked deformations (red dashed lines indicate the outer edges of cracks, openings and fractures in the internal structure of the point). The segmentation algorithm provides a 2.5D insight into the anatomical structure of the artefact after reconstruction. The outer surface boundaries of the artefact are marked in blue, with the light blue representing the inner openings, cracks and other deviations. The green colour represents the inner boundaries of the woody part of the artefact.

# 6.4. Comparison of the Quality of Surface-Based 3D Models of the Palaeolithic Wooden Point

To determine the efficiency of the algorithm and the quality of the reconstructed 3D surface-based model, we compared the 3D model of the wooden point reconstructed with the algorithms *dAR3D* and *sAR3D* with the 3D surface-based model of the point reconstructed with the commercial software package Avizo Fire (FEI). The comparison (graphical and statistical) was performed using the ICP algorithm and the CloudCompare graphical software tool. The comparison showed slight discrepancies between the two models. From this comparison it can be concluded that the quality of the reconstructed 3D model depends mainly on the quality of the input data. This fact also became clear when comparing two models that were created at different resolutions. Here the differences were in the contrast and texture details of the surface-based 3D model and not in its volumetric deviations.

# 6.5. Anatomical Characteristics of Surface-Based and Volumetric 3D Models of the Palaeolithic Wooden Point (2018–2019)

In the volumetric computer analysis of the anatomical structure of the volumetric 3D models, we restricted ourselves to a model reconstructed with a segmentation algorithm *sAR3D*. During segmentation, we paid attention to dislocations, inclusions, pores, cracks, openings, damages, crisps, deformations and fractures. We located critical areas, identified damage and changes that were previously not readable from the 3D surface-based model.

The critical points of the artefact's internal structure were determined volumetrically and marked on the model (Figures 15–18). Three pronounced internal deformations were detected (Figures 19 and 20): an elongated crack (B), a pronounced fracture (A) and a deformation (C). These three internal deformations stimulated the bending of the upper and lower part of the point after the conservation process, which was detected in the volumetric analysis and the comparison of the 3D surface models.



View from the planting part towards the top

**Figure 15.** Exposed critical points in a 3D volumetric model of a Palaeolithic wooden point. A bluegreen grid was chosen to make the anatomical structure of the artefact clearer. The light blue colour indicates the outer surface boundaries of the artefact and the inner boundaries of the non-wooden deformations (openings, fractures, cracks, pores, etc.) in the anatomical structure. The green colour indicates the inner boundaries of the wooden part of the artefact. The images show a view of the inner structure from the tip to the handle part (1' and 2') and from the handle part to the upper part of the artefact (1–4). Deviations and critical points are clearly visible.



**Figure 16.** Volumetric microlocations of critical sites in the volumetric 3D model of the Palaeolithic wooden point.



Figure 17. Overview of the critical points in the volumetric 3D model of the Palaeolithic wooden point.



**Figure 18.** Locations of exposed deformations of the Palaeolithic wooden point in the volumetric 3D model, which was recorded with a  $\mu$ CT scanner in 2019.

A comparison was made between models of the Palaeolithic point reconstructed in 2018 and 2019 using a CT scanner (Figures 21 and 22). Compared to the 2018 reconstructed model, the length of the point has shortened by 1.3 mm (-0.84%) in 1 year due to changes in deformation (shrinkage and bending of the point). The width and thickness at the most exposed points have increased—due to the expansion and deepening of the internal cracks—by 2.4 mm or 1.6% (width) and 1.1 mm or 0.73% (thickness), respectively. The internal dynamics of the surface changes show a tendency to settle down in this relatively short time interval. This fact underlines that the interval of the conservation process is longer in time than its official conclusion. The artefact is a "living organism" responding to altered ex situ nutritional conditions.



**Figure 19.** A three-dimensional depth image (viewed from the handle section) of the exposed critical areas in the anatomical structure of a Palaeolithic wooden point. Three main deformations were noted in the anatomical structure: a crack (B) running the entire length of the surface of the artefact; a transverse fracture (A) extending from the sampling point to the centre of the artefact; and numerous deformations, fractures, pores and cracks in the left wing of the artefact (C).



**Figure 20.** Fracture (A), which runs from the junction of the socket part and the point into the interior approx. 4.7 cm. A longer opening (B) is visible inside and cracks and fractures (C) in the left wing of the Palaeolithic wooden point.

A comparison of the anatomical changes of the artefact after one year shows a tendency towards greater contraction of the tip (Figures 23 and 24). This process also affects the condition of the longer crack running from the top towards the middle and to the plant part. The crack has widened (from 0.1 to 0.8 mm) and deepened further (from 0.1 to 1.4 mm) within a year. Deformation changes (widening of the cracks and smaller fractures—from 0.12 to 1.2 mm) were observed in the left wing of the artefact. The pronounced transverse fracture was stable over one year.

The deformation changes seen in Figure 24, which are not yet fully stabilised, indicate the hypothesis that the surface changes in the tip (bending, etc.) are mainly influenced by the crack that extends over the entire length in the upper part of the point. The dynamics of the changes at this crack have so far been faster than the dynamics of the changes at the lower fracture. A crack could be a major cause of bending of the plant part and the top of the point. In the long term, it could bring the risk of fracture or disintegration of the point.



**Figure 21.** Changes in surface deformation of the 2019 3D model of the Palaeolithic wooden point, compared to the 2018 reference model (changes are in a limited range between 0.0001 and 1.5001 mm). The colour matrix scale of deformation monitoring of the 2018 and 2019 3D models confirms the one-year dynamics of surface changes. Compared to other 3D models (2009–2017), the dynamics of changes on the artefact surface has stabilised. The shrinkage of the artefact persists with an average of 0.1 mm per year (green grid). However, the deformation of the uppermost point is even more pronounced. It lies between 0.5 mm and 1.0 mm (red-orange-yellow grid). The annual bending of the top by 1.35 mm is confirmed volumetrically. The bending is detected in the area of the handle. This has bent by 1.7 mm compared to 2018. More deformation of the left wing (A) of the point was also noted. In this area, the anatomical model drew attention to a number of unnatural internal cracks and deformations. Deformation variations in this area ranged from 1.1 mm to 2.2 mm compared to 2018, and the crack in the central part (B) widened by 1.4 mm. At the exposed point (B), it reaches a depth of 7.1 mm. Monitoring of the deformation was carried out using the CloudCompare software tool and the C2M algorithm (ICP).







**Figure 22.** Exposed sites of changes in the surface-based 3D model of the Palaeolithic wooden point (2019 —comparison with the 3D model from 2018). Volumetric measurements confirm the calming of the point deformation process. It is still dominated by shrinkage or bending in the range of 0.18—0.37 mm. Stand out (red value on the deformation scale—from 1.2 to 1.5 mm) deformation changes in the top, planting part and left wing point.

# 6.6. Reconstruction of 3D Models of other Archaeological Objects and Composite Materials

Experimental tests of 3D model reconstruction with our algorithms were conducted on four smaller archaeological objects made of bone and clay and two other materials (concrete and fabric) to determine the robustness and quality of the algorithms for 3D model reconstruction of surfaces and volumes (Figure 25).

In all cases, the two algorithms have been shown to be suitable for the reconstruction of 3D models. As for the quality of the 3D models reconstructed and created with the algorithms *dAR3D* and *sAR3D*, they do not differ in any way from the quality of the models reconstructed with commercial software tools. It is significant that the segmentation algorithm (Figure 25) matches and even surpasses the quality, clarity and robustness of the 3D models reconstructed with commercial software tools. At the same time, the reconstructed models provide accurate volumetric data on the location and condition at the critical points.



**Figure 23.** Annual dynamics of changes in anatomical structure (model—2019; reference model—2018)—view from the grip area to the top of the Palaeolithic wooden point. The deformation changes in the anatomical structure are clearly visible (crack along the entire length of the upper part of the artefact (1); larger fracture (2) in the lower and middle part; numerous unnatural deformations (3) in the left wing). The colour scale of the changes (red, green and orange grid) highlights the anatomical changes of the upper part of the artefact. The process of crack propagation and deformation of the tip is also shown.



**Figure 24.** Deformation monitoring of the top of the artefact (comparison of the 2019 model with the 2018 reference model). (**A**) shows the dynamics of volumetric changes on the surface, and (**B**) shows the dynamics of changes in the anatomical structure of the artefact. The colour scale represents the annual process of shrinkage and deformation of the upper part of the artefact. The dynamics range from 0.1 mm (green) to 1.1 mm (red). The upper side of the artefact is mainly exposed to a more intensive deformation process. On the inside, smaller cracks, openings and pores can be seen. These touch the beginning of the crack, which extends over the entire length from the top to the handle area. Due to the ongoing deformation process in this part of the artefact, the annual deflection of the top of the Palaeolithic head in 2019 was 1.35 mm.



**Figure 25.** Six examples of reconstruction of 3D models from CT images of various archaeological objects and other composite materials. On the left are three clay rattles [93], one the right a Neanderthal bone whistle [69] (top), a cylindrical piece of concrete (middle) and some fabric (bottom).

#### 7. Discussion

The iterative segmentation algorithm developed for reconstructing a 3D model from 2D micro-CT images of an archaeological object is only one of a number of current algorithms for processing 2D CT images or 3D CT sections. However, it is one of the first to use a computer vision approach to reconstruct 3D surface and anatomical models and addresses the weaknesses of traditional techniques for reconstructing 3D surface models from 2D images that occur when using conventional computer vision algorithms. It combines knowledge, methods and techniques from computed tomography and computer vision. It provides extremely accurate 3D surface and anatomical models that, together with high-quality CT images, can be a perfect copy of the original. Compared to other algorithms, it is simple and user-friendly. It is versatile and not specialised for narrow professional use (although it was developed for direct use by archaeologists and conservators). With reconstructed 3D models, it provides the user with a large amount of data and information and enriches archaeological documentation. The image data and information can be processed with already-available software tools for processing 3D network data. However, it is true that this makes the reconstruction of 3D models more expensive than traditional computer vision algorithms. With the financial availability of CT hardware, it is also becoming interesting for use in the heritage sciences. This is confirmed by the growing number of research projects and published professional contributions in all fields and not only in heritage studies.

With the developed segmentation algorithm we used in our specific archaeological case, the process of segmenting features in 2D images using the Roberts Edge detector, after reconstructing the 3D model (from any plane section) we obtain a depth image of the object in the case of a dynamic 3D visualisation or a printed 2.5-D image simulating the holographic method of recording 3D dimensional information about an archaeological object.

Our simplification of the preprocessing of 2D image files or 3D slices eliminates the need to use additional philtres (Rashidi, Vigorelli, Bakirov) for the segmentation process. It

is also not necessary to use the translucent visualisation technique (Gaboutchian) of the 3D anatomical model. The depth view (a 2.5D depth view and not just a surface hologram of the outer edges of selected internal segmented features—see Figures 12, 13, 17, 21, 22, 23) was not provided by any software tool for reconstructing 3D anatomical models from CT images. Anatomical views are therefore simulated in most cases by visualising only the outer edges of selected internal/anatomical deviations or features. To achieve a similar effect in visualising anatomical features—but without insight into the interior of the anatomical feature—a simple video sequence of 2D images of the reconstructed model in the selected plane is usually used to analyse the anatomical features of the 3D model. In this case, it is left to the user to visualise and localise the features and spatial properties of the selected anatomical feature by numerically recording metric data about its location, which is automatically acquired by the software tool. The volumetric correctness of the placement of the reconstructed 3D model in the Cartesian coordinate system may also be questionable. It may deviate from the original. Therefore, volumetric deviations and errors occur more frequently. This may complicate the volumetric analysis of the artefact or the design of conservation procedures. Especially when machine or deep learning, artificial intelligence or automatic or robotic processing methods or techniques are involved in the processes of analysis or conservation.

We do not exclude another way of visualising the anatomical structure of the object, but after reviewing the software tools of the user manual, such a possibility is neither described nor foreseen.

The highlighted features of the segmentation algorithm require critical consideration and implementation in new cases and in different practises to objectively assess its applicability. Our evaluation has shown that the new algorithms for reconstructing 3D models from 2D images CT have met expectations. Critical points in the internal structure of the Palaeolithic wooden point, which was the subject of our investigation, were successfully located. Anatomical features point to the causes of the surface changes. The need for further development of better conservation techniques for wet wood is highlighted.

In all cases, the algorithms qualitatively reconstructed the surface-based 3D models and the volumetric 3D models from CT images. The reconstructed 3D models do not differ from the quality of the models reconstructed or rendered using commercial software tools. This is an indication that the segmentation algorithm *sAR3D* reconstructs models that outperform those reconstructed with commercial software tools in terms of quality, clarity and volumetric accuracy.

The developed iterative algorithm for reconstructing 3D models from 2D microtomographic images of archaeological objects can be successfully used to create both surfacebased and volumetric 3D models of X-ray scanned objects. They also enable the reconstruction of a three-dimensional 3D model after previously constraining the number of features specified by the user.

The algorithms were tested on seven practical examples and met expectations. The results answered the working questions (reconstruction of 3D models from 2D  $\mu$ CT images, internal deformation of the Palaeolithic artefact).

With the reconstructed 3D models, we were able to accurately identify, examine and document the internal structure of the artefact. Deformations (cracks, fractures, decay) are clearly visible and volumetrically localised. The 3D anatomical model successfully complemented the knowledge of the features and specificities of the alterations and revised the erroneous and not fact-based assessment of the alterations that was solely based on the surface-based 3D models.

The chosen hardware and software proved to be suitable to realise the set development and assessment goals of the study. It can form a basis for a possible standardisation of hardware and software for the reconstruction of computed tomography images into a 3D model.

The advantage of both algorithms is their ease of use. The user can reconstruct the 3D model according to his research objectives, with the appropriate hardware and open-

source software configuration of the computer system, without any special mathematical or computer knowledge and without additional professional and technical support from a radiologist, computer scientist or physicist.

To implement the algorithm directly, the user needs:

- the selected number of CT images in the appropriate format (the set of images for reconstruction is not limited in number; in our case, for example, the set of CT images in the selected test objects ranged from 1100 to 3500 images);
- data and information from the radiologist about the thickness of the layer (d = t = z) and the type of X-ray beam;
- information about the possible inclination (in degrees) of the object mounted in the CT reader, and;
- information from the radiologist about the limits of the object (length width thickness) that can be detected by the selected CT reader with a single X-ray (scan).

The information and data provided are important for the accurate volumetric treatment of the object and its match to the original. Algorithms otherwise adapted for the use of computed tomography and 3D model reconstruction from CT images of small archaeological objects for archaeological documentation and for planning conservation and restoration procedures can also be robustly used in other cases (e.g., testing and analysis of materials or composites, in industrial control, etc.). The limit for the choice of the algorithm's intended use depends on the technical possibilities of the chosen CT reader.

The algorithms are useful in the reconstruction of 3D models from 2D microtomographic or tomographic images acquired with all types of CT or  $\mu$ CT readers, as well as with other depth imaging readers (e.g., MRI, etc.). The segmentation algorithm can be fully adapted to the needs of different users and workspaces. It is simple and straightforward and does not require in-depth mathematical and computer knowledge of the direct user.

The quality of the reconstructed anatomical (3D volumetric) and surface-based 3D models from 2D  $\mu$ CT images is not significantly different from the quality of 3D models created with more expensive commercial software. The evaluation of the algorithms confirmed that the quality of the 3D model depends mainly on the quality of the input 2D CT or  $\mu$ CT images.

Given the amount of information contained in the internal texture of 2D  $\mu$ CT images, the *dAR3D* algorithm is significantly slower than the *sAR3D* in reconstructing the 3D model. The shorter reconstruction time of the 3D model with the *sAR3D* algorithm is due to the smaller amount of information in the segmented 2D $\mu$ CT images. The difference between the two algorithms also lies in the size of the output file of the 3D model. The *dAR3D* file is significantly larger (Table 6).

It is critical to note that the current technological requirements for an optimal and rapid open-source implementation of the direct algorithm in the context of the use of personal computers for general use are not yet fully in place. With a large number of images (more than 1000) with high resolution ( $3000 \times 3000$  or more) and a high density of greyscale values of the HU or RGB matrix, the reconstruction process with a direct algorithm can be very time consuming. These circumstances significantly extend the time limits for effective reconstruction of 3D models, but they can be overcome with the help of cloud processing (4th data processing paradigm—op. Cit.) [94–96].

Advantages, disadvantages and limitations of both algorithms are summarised in Table 7.

The reconstruction of surface-based and volumetric 3D models of an archaeological object with both algorithms confirmed the extraordinary informative value of the model for an objective assessment of the current state of the artefact. The additional information about the condition of the artefact obtained in this way can also be of great help to conservators in selecting the most appropriate methods and techniques for conservation and protection.

Artefacts	3D Model—Reconstruction Time and Output File Size				
	dAR3D	File Size	Format	sAR3D	File Size
Palaeolithic wooden point	48 h (year 2018) 36 h (year 2019)	8.18 GB 7.7 GB	OBJ	1.10 h (year 2018) 1.04 h (year 2019)	196 MB 193 MB
Bone flute from Divje Babe I	24 h	4.68 GB	OBJ	55′	166 MB
Ceramic rattles	R1—23.9 h R2—16.0 h R3—11.7 h	5.12 GB 3.26 GB 2.49 GB	OBJ	R1—45' R2—30' R3—24'	132.0 MB 89.2 MB 68.1 MB
Different composites	CONCRETE—18.1 h FABRIC—8.7 h	14.9 GB 1.8 GB	OBJ	CONCRETE—78' FABRIC—20'	280 MB 69 MB

**Table 6.** Reconstruction time and output file size of algorithms *dAR3D* and *sAR3D*.

 Table 7. Advantages, limitations and deficiencies of algorithms dAR3D and sAR3D.

	dAR3D	sAR3D
Advantages	<ul> <li>Complete surface and volume reconstruction of the 3D model; suitable for quality and complete addition of the original.</li> <li>Suitable for the reconstruction of small artefacts.</li> <li>Suitable for the reconstruction of up to 300 CT, MRI, ultrasound, MMG 2D images.</li> </ul>	<ul> <li>Fast, reliable and efficient reconstruction of the 3D model.</li> <li>High-quality and accurate surface 3D model for visualisation and addition.</li> <li>High-quality and more segmented 3D volume model according to selected characteristics.</li> <li>More vivid and selective presentation and analysis of 3D model data.</li> <li>Adaptation to the interests and needs of the end user.</li> <li>Simple and easy by the end user.</li> <li>Robustness (can be used in various fields).</li> <li>Suitable for processing and processing a large number of CT, μCT, then- CT,, MRI, ultrasound, MMG, 2D images (1000 <n).< li=""> <li>Ability to remove unwanted data.</li> <li>Lower memory and hardware load.</li> <li>Efficient and fast operation regardless of the number of 2D images reconstructed.</li> <li>Fast and efficient comparison and processing of volumetric data from surface and volume 3D model.</li> <li>Efficient and fast implementation of deformation monitoring.</li> <li>Smaller and more suitable file of reconstructed 3D models for further processing.</li> </n).<></li></ul>
Limitations and Deficiencies	<ul> <li>Longer time intervals of 3D model reconstruction from µCT images (t = 25–50 x; depending on architecture and hardware capabilities).</li> <li>Optimal processing in the range of up to 300 2D images.</li> <li>Extremely large files when reconstructing from a larger set of 2D images (over 1000) with a higher resolution (e.g., 15–50 GB).</li> <li>Increased saturation and therefore the risk of noise and poorer contrast.</li> <li>Poorer quality (contrast) due to a larger set of greyscales HU or RGB scales.</li> <li>Lots of useless and unstructured data.</li> </ul>	

# 8. Conclusions

Professional work in the field of cultural heritage conservation, archaeology, conservation and museology cannot be imagined without the use of new scientific methods and techniques originating in the natural sciences, or without the use of new computer and information technologies, artificial intelligence, collections of 3D models, multimedia, robotics, computer analysis of deformations and other modern technologies (Visual Languages, Remote Sensing, AI Neural Networks, Big Data, Data Science, etc.). Indeed, 3D surface-based models and volumetric 3D models (anatomical), volumetric 3D analyses and CT have become indispensable for documentation for the permanent, integrated conservation and presentation of tangible cultural heritage today and in the future.

The importance of 3D surface-based and volumetric 3D models and 3D computer visualisations enriches the standards for archaeological and cultural heritage recommended by the London Charter [10], the Seville Principles [11] and ratified international treaties.

At the same time, computed tomography and the reconstruction of 3D volumetric (anatomical) models and 3D surface-based models from tomographic or microtomographic images can effectively contribute to the implementation of Rule 4 of the Amendments to the UNESCO Convention for the Protection of the Underwater Cultural Heritage, which recommends that state authorities and institutions apply nondestructive methods to cultural heritage remains [6].

It would be appropriate for the archaeological and conservation professions to use noninvasive computed tomography more frequently than in the past and to model successful practises as standard methods and techniques for dealing with fragile and rapidly degradable cultural heritage remains.

Both the algorithms for reconstructing 3D surface-based and 3D volumetric models from CT images can be upgraded with new methods from the fields of computer vision, deep learning and artificial intelligence. By upgrading algorithms to reconstruct 3D models, we could complement or even replace the autonomous analytical function of humans in the treatment and evaluation of cultural heritage remains. Experiences with algorithms such as Deep Learning already represent a significant advance in the automatic recognition and processing of 2D images from computed tomography. Methods and techniques ranging from convolutional neural networks (CNNs) to variational auto-encoders (VAEs) are being used. These artificial intelligence methods are already delivering successful results in the automatic recognition of complex patterns in computed tomographic image data, particularly in the faster and more efficient segmentation of features and the automatic diagnosis of X-ray image conditions [97].

Author Contributions: Conceptualization, F.S., M.E. and E.G.P.; CT scanning, L.K.; methodology, F.S., M.E., L.K. and E.G.P.; software, E.G.P.; resources, E.G.P., F.S., A.J. and M.E.; data curation, E.G.P. and L.K.; investigation, E.G.P; M.E.; L.K. and F.S.; algorithm development and simulation, E.G.P.; validation, F.S., M.E. and L.K.; formal analysis, E.G.P.; M.E. and L.K.; supervision, F.S. and A.J.; writing—original draft preparation, E.G.P.; writing—review and editing, F.S., M.E., A.J. and L.K.; All authors have read and agreed to the published version of the manuscript

Funding: This research was supported by The Slovenian Research Agency (ARRS), grant P2-0273.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Acknowledgments: The authors would like to thank The Archaeological Department of the National Museum of Slovenia and curator Peter Turk for permission to use the digital CT documentation of archaeological objects to evaluate and test the robustness of new algorithms on archaeological objects.

Conflicts of Interest: The authors declare no conflict of interest.

## Abbreviations

The following abbreviations are used in this manuscript:

СТ	computed tomography	
dAR3D	direct algorithm for 3D model reconstruction	
sAR3D	segmentation algorithm for 3D model reconstruction	

# References

- Guček Puhar, E.; Korat, L.; Erič, M.; Jaklič, A.; Solina, F. Reconstruction of 3D models from microtomographic images of archeological artifacts. In Proceedings of the 2020 IMEKO TC-4 International Conference on Metrology for Archaeology and Cultural Heritage, Trento, Italy, 22–24 October 2020; pp. 378–383.
- Gaspari, A.; Erič, M.; Odar, B. A Palaeolithic wooden point from Ljubljansko barje, Slovenia. Submerged Prehistory; Benjamin, J.; Bonsall, C.; Pickard, C.; Fischer, A., Eds.; Oxbow Books: Oxford, UK, 2011; pp. 186–192. [CrossRef]
- Erič, M.; Gaspari, A.; Čufar, K.; Solina, F.; Verbič, T. Zgodnjerimska ladja iz Ljubljanice pri Sinji Gorici = early Roman barge from the Ljubljanica river at Sinja Gorica. Arheološki Vestnik 2014, 65, 187–254.
- Erič, M.; Guček Puhar, E.; Jaklič, A.; Solina, F. The Necessity of Changing the Methodology of Preserving Waterlogged Wooden Objects. Skyllis Z. Für Marit. Und Limnische Archäologie Und Kult.gesch. 2018, 2018, 174–185.
- Guček Puhar, E.; Erič, M.; Kavkler, K.; Celec, K.; Cramer, A.; Korat, L.; Jaklič, A.; Solina, F. Comparison and deformation analysis of five 3D models of the Paleolithic wooden point from the Ljubljanica River. In Proceedings of the 2018 Metrology for Archaeology and Cultural Heritage (MetroArchaeo), Cassino, Italy, 22–24 October 2018; pp. 449–454. [CrossRef]
- 6. Erič, M.; Stopar, D.; Solina, F.; Kavkler, K. Reconceptualization of the contemporary maritime museum. Do we really need the original waterlogged wooden artefacts and objects? *Skyllis Z. Für Marit. Und Limnische Archäologie Und Kult.gesch.* **2019**, *19*, 11–28.
- Solina, F. 3D Point Clouds for Documentation, Analysis and Promotion of Cultural Heritage Artifacts. In Proceedings of the 23rd International Conference on Cultural Heritage and New Technologies, Vienna, Austria, 12–15 November 2018; pp. 1–9.
- 8. Jaklič, A.; Leonardis, A.; Solina, F. *Segmentation and Recovery of Superquadrics*; Computational Imaging and Vision; Kluwer Academic Publishers: Dordrecht, The Nethereland, 2000; Volume 20. [CrossRef]
- Jaklič, A.; Erič, M.; Mihajlović, I.; Stopinšek, Ž.; Solina, F. Volumetric models from 3D point clouds: The case study of sarcophagi cargo from a 2nd/3rd century AD Roman shipwreck near Sutivan on island Brač, Croatia. J. Archaeol. Sci. 2015, 62, 143–152. [CrossRef]
- 10. Denard, H. A new introduction to the London Charter. In *Paradata and Transparency in Virtual Heritage*; Routledge: London, UK, 2016; pp. 83–98.
- López-Menchero Bendicho, V.M. International Guidelines for Virtual Archaeology: The Seville Principles. In *Good Practice in Archaeological Diagnostics. Natural Science in Archaeology;* Corsi, C., Slapšak, B., Vermeulen, F., Eds. Springer: Cham, 2013. [CrossRef]
- 12. Cieslik, E. 3D Digitization in Cultural Heritage Institutions Guidebook. Available online: https://www.dental.umaryland.edu/ media/sod/national-museum-of-dentistry/pdfs/3D-Digitization-Guidebook.pdf (accessed on 10 October 2021).
- Mathys, A.; Lemaitre, S.; Brecko, J.; Semal, P. Agora 3D: Evaluating 3D Imaging Technology for the Research, Conservation and Display of Museum Collections. *Antiq. Proj. Gallery* 2013, 87. Available online: http://antiquity.ac.uk/projgall/mathys336/ (accessed on 10 October 2021).
- 14. Schwab, K. The Fourth Industrial Revolution; Crown Business: New York, NY, USA, 2017.
- 15. Flitton, G.; Mouton, A.; Breckon, T.P. Object classification in 3D baggage security computed tomography imagery using visual codebooks. *Pattern Recognit.* 2015, 48, 2489–2499. [CrossRef]
- 16. Cnudde, V.; Boone, M.N. High-resolution X-ray computed tomography in geosciences: A review of the current technology and applications. *Earth-Sci. Rev.* 2013, 123, 1–17. [CrossRef]
- 17. Liu, L. Model-based iterative reconstruction: A promising algorithm for today's computed tomography imaging. *J. Med Imaging Radiat. Sci.* **2014**, 45, 131–136. [CrossRef]
- Elmokadem, A.H.; Ibrahim, E.A.; Gouda, W.A.; Razek, A.A.K.A. Whole-Body Computed Tomography Using Low-Dose Biphasic Injection Protocol With Adaptive Statistical Iterative Reconstruction V: Assessment of Dose Reduction and Image Quality in Trauma Patients. J. Comput. Assist. Tomogr. 2019, 43, 870–876. [CrossRef]
- 19. Case, J.T.; Kenderian, S.; Johnson, E.C. Portable X-ray Computed Tomography. US Patent 10,096,148, 9 October 2018.
- 20. Grangeat, P.; Amans, J.L. *Three-Dimensional Image Reconstruction in Radiology and Nuclear Medicine*; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2013; Volume 4. [CrossRef]
- 21. Dan, P.T.; Porat, R.; Kalisman, O. Systems and Methods for an Integrated System for Visualizing, Simulating, Modifying and 3D Printing 3D Objects. US Patent 10,275,909, 30 April 2019.
- Kulczyk, T.; Rychlik, M.; Lorkiewicz-Muszyńska, D.; Abreu-Głowacka, M.; Czajka-Jakubowska, A.; Przystańska, A. Computed Tomography versus Optical Scanning: A Comparison of Different Methods of 3D Data Acquisition for Tooth Replication. *BioMed Res. Int.* 2019, 2019, 4985121. [CrossRef]
- 23. Jinsihan, N.; Jin, G. Recent developments of 3D-printing technique assisted surgery in the management of complex fractures. *Int. J. Clin. Exp. Med.* **2018**, *11*, 11578–11583.
- 24. Hughes, S. Three-dimensional reconstruction of an ancient Egyptian mummy. In *Imaging the Past. Electronic Imaging and Computer Graphics in Museums and Archaeology*; Higgins, T., Main, P., Lang, J., Eds.; The British Museum: London, UK, 1996; Volume 114, pp. 211–225.

- Murphy Jr, W.A.; Nedden, D.z.; Gostner, P.; Knapp, R.; Recheis, W.; Seidler, H. The iceman: Discovery and imaging. *Radiology* 2003, 226, 614–629. [CrossRef]
- Weber, G.W.; Lukeneder, A.; Harzhauser, M.; Mitteroecker, P.; Wurm, L.; Hollaus, L.M.; Kainz, S.; Haack, F.; Antl-Weiser, W.; Kern, A. The microstructure and the origin of the Venus from Willendorf. *Sci. Rep.* 2022, *12*, 1–10. [CrossRef]
- Applbaum, N.; Applbaum, Y. The use of medical computed tomography (CT) imaging in the study of ceramic and clay archaeological artifacts from the ancient near east. In *X-rays for Archaeology*; Uda, I., Demortler, G., Nakai, I., Eds.; Springer: Berlin/Heidelberg, Germany, 2005; pp. 231–245. [CrossRef]
- Rossi, M.; Casali, F.; Chirco, P.; Morigi, M.; Nava, E.; Querzola, E.; Zanarini, M. X-ray 3D computed tomography of bronze archaeological samples. *IEEE Trans. Nucl. Sci.* 1999, 46, 897–903. [CrossRef]
- 29. Re, A.; Corsi, J.; Demmelbauer, M.; Martini, M.; Mila, G.; Ricci, C. X-ray tomography of a soil block: A useful tool for the restoration of archaeological finds. *Herit. Sci.* 2015, *3*, 4. [CrossRef]
- Albertin, F.; Bettuzzi, M.; Brancaccio, R.; Morigi, M.P.; Casali, F. X-Ray Computed Tomography In Situ: An Opportunity for Museums and Restoration Laboratories. *Heritage* 2019, 2, 2028–2038. [CrossRef]
- Vigorelli, L.; Re, A.; Guidorzi, L.; Cavaleri, T.; Buscaglia, P.; Nervo, M.; Facchetti, F.; Borla, M.; Grassini, S.; Lo Giudice, A. X-ray Imaging Investigation on the Gilding Technique of an Ancient Egyptian Taweret Wooden Statuette. *J. Imaging* 2021, 7, 229. [CrossRef]
- 32. Rashidi, A.; Olfatbakhsh, T.; Crawford, B.; Milani, A.S. A Review of Current Challenges and Case Study toward Optimizing Micro-Computed X-Ray Tomography of Carbon Fabric Composites. *Materials* **2020**, *13*, 3606. [CrossRef]
- Gaboutchian, A.V.; Knyaz, V.A.; Korost, D.V. New Approach to Dental Morphometric Research Based on 3D Imaging Techniques. J. Imaging 2021, 7, 184. [CrossRef]
- 34. Bakirov, B.; Saprykina, I.; Kichanov, S.; Mimokhod, R.; Sudarev, N.; Kozlenko, D. Phase Composition and Its Spatial Distribution in Antique Copper Coins: Neutron Tomography and Diffraction Studies. *J. Imaging* **2021**, *7*, 129. [CrossRef]
- 35. Takenouchi, K.; Yamahana, K. Fine pottery shaping techniques in Predynastic Egypt: A pilot study on non-destructive analysis using an X-Ray CT scanning system. *J. Archaeol. Sci. Rep.* **2021**, *37*, 102989. [CrossRef]
- Żyła, K.; Kęsik, J.; Santos, F.; House, G. Scanning of Historical Clothes Using 3D Scanners: Comparison of Goals, Tools, and Methods. *Appl. Sci.* 2021, 11, 5588. [CrossRef]
- 37. Brunello, V.; Canevali, C.; Corti, C.; De Kock, T.; Rampazzi, L.; Recchia, S.; Sansonetti, A.; Tedeschi, C.; Cnudde, V. Understanding the Microstructure of Mortars for Cultural Heritage Using X-ray CT and MIP. *Materials* **2021**, *14*, 5939. [CrossRef] [PubMed]
- Ricci, M.; Laureti, S.; Malekmohammadi, H.; Sfarra, S.; Lanteri, L.; Colantonio, C.; Calabrò, G.; Pelosi, C. Surface and Interface Investigation of a 15th Century Wall Painting Using Multispectral Imaging and Pulse-Compression Infrared Thermography. *Coatings* 2021, 11, 546. [CrossRef]
- Rode, S.; Singh, H.; Thube, N.; Mhaske, M.; Ansari, H.; Vadhel, M. Design and Fabrication of a 3-D scanning system using Optical Computed Tomography and Photogrammetry. In Proceedings of the 2021 4th Biennial International Conference on Nascent Technologies in Engineering (ICNTE), Navi Mumbai, India, 15–16 January 2021; pp. 1–5.
- 40. Marr, D.; Ullman, S.; Poggio, T.U. Vision: A Computational Investigation into the Human Representation and Processing of Visual Information; MIT Press: Cambridge, MA, USA, 2010. [CrossRef]
- 41. Bajcsy, R.; Aloimonos, Y.; Tsotsos, J.K. Revisiting active perception. Auton. Robot. 2018, 42, 177–196. [CrossRef]
- 42. Lanfranco, A.R.; Castellanos, A.E.; Desai, J.P.; Meyers, W.C. Robotic surgery: A current perspective. *Ann. Surg.* 2004, 239, 14. [CrossRef]
- 43. Le Moal, J.; Peillon, C.; Dacher, J.N.; Baste, J.M. Three-dimensional computed tomography reconstruction for operative planning in robotic segmentectomy: A pilot study. *J. Thorac. Dis.* **2018**, *10*, 196. [CrossRef]
- Zhou, M.; Hamad, M.; Weiss, J.; Eslami, A.; Huang, K.; Maier, M.; Lohmann, C.P.; Navab, N.; Knoll, A.; Nasseri, M.A. Towards robotic eye surgery: Marker-free, online hand-eye calibration using optical coherence tomography images. *IEEE Robot. Autom. Lett.* 2018, *3*, 3944–3951. [CrossRef]
- Mastrodicasa, D.; Albrecht, M.H.; Schoepf, U.J.; Varga-Szemes, A.; Jacobs, B.E.; Gassenmaier, S.; De Santis, D.; Eid, M.H.; van Assen, M.; Tesche, C.; et al. Artificial intelligence machine learning-based coronary CT fractional flow reserve (CT-FFRML): Impact of iterative and filtered back projection reconstruction techniques. J. Cardiovasc. Comput. Tomogr. 2019, 13, 331–335. [CrossRef]
- 46. Rinaldi, R.G.; Blacklock, M.; Bale, H.; Begley, M.R.; Cox, B.N. Generating virtual textile composite specimens using statistical data from micro-computed tomography: 3D tow representations. *J. Mech. Phys. Solids* **2012**, *60*, 1561–1581. [CrossRef]
- 47. Mo, K.; Wang, H.; Yan, X.; Guibas, L.J. PT2PC: Learning to Generate 3D Point Cloud Shapes from Part Tree Conditions. *arXiv Prepr.* **2020**, arXiv:2003.08624.
- 48. Czabaj, M.W.; Riccio, M.L.; Whitacre, W.W. Numerical reconstruction of graphite/epoxy composite microstructure based on sub-micron resolution X-ray computed tomography. *Compos. Sci. Technol.* **2014**, *105*, 174–182. [CrossRef]
- Pan, Z.; Tian, S.; Guo, M.; Zhang, J.; Yu, N.; Xin, Y. Comparison of medical image 3D reconstruction rendering methods for robot-assisted surgery. In Proceedings of the 2017 2nd International Conference on Advanced Robotics and Mechatronics (ICARM), Hefei, China, 27–31 August 2017; pp. 94–99. [CrossRef]
- 50. Pawasauskas, J. Volume visualization with ray casting. Adv. Top. Comput. Graph. 1997.

- 51. Kalshetti, P.; Rahangdale, P.; Jangra, D.; Bundele, M.; Chattopadhyay, C. Antara: An Interactive 3D Volume Rendering and Visualization Framework. *arXiv Prepr.* **2018**, arXiv:1812.04233.
- 52. Drebin, R.A.; Carpenter, L.; Hanrahan, P. Volume rendering. ACM Siggraph Comput. Graph. 1988, 22, 65–74. [CrossRef]
- 53. LaRose, D.A. Iterative X-ray/CT Registration Using Accelerated Volume Rendering. Ph.D. Thesis, Carnegie Mellon University, Pittsburgh, PA, USA, 2001.
- 54. Fortmeier, D. Direct Volume Rendering Methods for Needle Insertion Simulation. Ph.D. Thesis, University of Lübeck, Lubeck, Germany, 2016.
- 55. Csebfalvi, B.; Szirmay-Kalos, L. Monte carlo volume rendering. In Proceedings of the IEEE Visualization, Seattle, WA, USA, 19–24 October 2003; pp. 449–456. [CrossRef]
- 56. Calhoun, P.S.; Kuszyk, B.S.; Heath, D.G.; Carley, J.C.; Fishman, E.K. Three-dimensional volume rendering of spiral CT data: Theory and method. *Radiographics* **1999**, *19*, 745–764. [CrossRef] [PubMed]
- 57. Shaw, C.C. (Ed.) Cone Beam Computed Tomography, 1st ed.; CRC Press: Boca Raton, FL, USA, 2014. [CrossRef]
- 58. Herman, G.T. *Fundamentals of Computerized Tomography: Image Reconstruction from Projections;* Springer Science & Business Media: Berlin/Heidelberg, Germany, 2009. [CrossRef]
- Carvalho, L.; Sobieranski, A.C.; von Wangenheim, A. 3D segmentation algorithms for computerized tomographic imaging: A systematic literature review. J. Digit. Imaging 2018, 31, 799–850. [CrossRef]
- Wan, X.; Yang, F.; Yang, F.; Li, X.; Xu, M.; Tian, J. Visualization of multiple anatomical structures with explicit isosurface manipulation. In Proceedings of the 2015 37th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), Milan, Italy , 25–29 August 2015; pp. 4234–4237. [CrossRef]
- 61. Geyer, L.L.; Schoepf, U.J.; Meinel, F.G.; Nance, J.W.; Bastarrika, G.; Leipsic, J.A.; Paul, N.S.; Rengo, M.; Laghi, A.; De Cecco, C.N. State of the Art: Iterative CT Reconstruction Techniques. *Radiology* **2015**, *276*, 339–357. [CrossRef]
- 62. Yasaka, K.; Kamiya, K.; Irie, R.; Maeda, E.; Sato, J.; Ohtomo, K. Metal artefact reduction for patients with metallic dental fillings in helical neck computed tomography: Comparison of adaptive iterative dose reduction 3D (AIDR 3D), forward-projected model-based iterative reconstruction solution (FIRST) and AIDR 3D with single-energy metal artefact reduction (SEMAR). *Dentomaxillofacial Radiol.* 2016, 45, 20160114. [CrossRef]
- 63. Ghetti, C.; Ortenzia, O.; Serreli, G. CT iterative reconstruction in image space: A phantom study. *Phys. Med.* **2012**, *28*, 161–165. [CrossRef]
- 64. Tipnis, S.; Ramachandra, A.; Huda, W.; Hardie, A.; Schoepf, J.; Costello, P.; Flohr, T.; Sedlmair, M. Iterative reconstruction in image space (IRIS) and lesion detection in abdominal CT. In Proceedings of the Proceedings Volume 7622, Medical Imaging 2010: Physics of Medical Imaging, San Diego, CA, USA, 13–18 February 2010; Volume 7622, p. 76222. [CrossRef]
- Schindler, A.; Vliegenthart, R.; Schoepf, U.J.; Blanke, P.; Ebersberger, U.; Cho, Y.J.; Allmendinger, T.; Vogt, S.; Raupach, R.; Fink, C.; et al. Iterative image reconstruction techniques for CT coronary artery calcium quantification: Comparison with traditional filtered back projection in vitro and in vivo. *Radiology* 2014, 270, 387–393. [CrossRef]
- Leipsic, J.; Labounty, T.M.; Heilbron, B.; Min, J.K.; Mancini, G.J.; Lin, F.Y.; Taylor, C.; Dunning, A.; Earls, J.P. Adaptive statistical iterative reconstruction: Assessment of image noise and image quality in coronary CT angiography. *Am. J. Roentgenol.* 2010, 195, 649–654. [CrossRef] [PubMed]
- 67. Dalehaug, I.; Bolstad, K.N.; Aadnevik, D.; Flataboe, S.; Pettersen, H.E.S. Admire vs. Safire: Objective comparison of CT reconstruction algorithms and their noise properties. *arXiv* 2017, arXiv:1708.09616.
- Fornaciari, G.; Bagnoli, J.; Minozzi, S.; Giuffra, V.; Paribeni, E.; Ferrari, V.; Caramella, D.; Mosca, F. Spiral Computed Tomography (CT) and ENDOCAS application in Archaeology to Etruscan Funerary Urns. In Proceedings of the 2012 Computer Assisted Radiology and Surgery (CARS), Pisa, Italy, 27–30 June 2012; Volume 7, pp. S508–S509.
- 69. Turk, M.; Turk, I.; Dimkaroski, L.; Blackwell, B.A.; Horusitzky, F.Z.; Otte, M.; Bastiani, G.; Korat, L. The Mousterian musical instrument from the Divje babe I cave (Slovenia): Arguments on the material evidence for Neanderthal musical behaviour. *L'anthropologie* **2018**, *122*, 679–706. [CrossRef]
- 70. Zur Nedden, D.; Knapp, R.; Wicke, K.; Judmaier, W.; Murphy Jr, W.A.; Seidler, H.; Platzer, W. Skull of a 5,300-year-old mummy: Reproduction and investigation with CT-guided stereolithography. *Radiology* **1994**, *193*, 269–272. [CrossRef]
- Laurinen, P.; Siirtola, P.; Röning, J. Efficient Algorithm for Calculating Similarity between Trajectories Containing an Increasing Dimension. In Proceedings of the IASTED International Conference on Artificial Intelligence and Applications, part of the 24th Multi-Conference on Applied Informatics, Innsbruck, Austria, 13–16 February 2006; pp. 392–399.
- 72. Verhoeven, G.; Doneus, M.; Briese, C.; Vermeulen, F. Mapping by matching: A computer vision-based approach to fast and accurate georeferencing of archaeological aerial photographs. *J. Archaeol. Sci.* **2012**, *39*, 2060–2070. [CrossRef]
- 73. Kersten, T.P.; Lindstaedt, M. Automatic 3D object reconstruction from multiple images for architectural, cultural heritage and archaeological applications using open-source software and web services. *Photogrammetrie* **2012**, 2012, 727–740. [CrossRef]
- 74. Tong, L.; Ying, X. 3D point cloud initial registration using surface curvature and SURF matching. 3D Res. 2018, 9, 41. [CrossRef]
- Wang, C.; Jiang, H. SURF: A connectivity-based space filling curve construction algorithm in high genus 3D surface WSNs. In Proceedings of the IEEE Conference on Computer Communications (INFOCOM), Hong Kong, China, 26 April–1 May 2015; pp. 981–989. [CrossRef]
- 76. Brutto, M.L.; Meli, P. Computer vision tools for 3D modelling in archaeology. Int. J. Herit. Digit. Era 2012, 1, 1–6. [CrossRef]

- 77. Forte, M.; Dell'Unto, N.; Issavi, J.; Onsurez, L.; Lercari, N. 3D archaeology at Çatalhöyük. *Int. J. Herit. Digit. Era* **2012**, *1*, 351–378. [CrossRef]
- Fassold, H.; Danzl, R.; Schindler, K.; Bischof, H. Reconstruction of archaeological finds using shape from stereo and shape from shading. In Proceedings of the 9th Computer Vision Winter Workshop, Piran, Slovenia, 4–6 February 2004; pp. 21–30.
- Doneus, M.; Verhoeven, G.; Fera, M.; Briese, C.; Kucera, M.; Neubauer, W. From deposit to point cloud–a study of low-cost computer vision approaches for the straightforward documentation of archaeological excavations. *Geoinform. FCE CTU* 2011, 6, 81–88. [CrossRef]
- Stamos, I.; Allen, P.K. Geometry and texture recovery of scenes of large scale. *Comput. Vis. Image Underst.* 2002, 88, 94–118. [CrossRef]
- Guo, J.; Wang, Y.; Chen, J.; Lin, J.; Wu, L.; Xue, K.; Liu, W.; Zhang, J. Markerless tracking for augmented reality applied in reconstruction of Yuanmingyuan archaeological site. In Proceedings of the 11th IEEE International Conference on Computer-Aided Design and Computer Graphics, Yellow Mountain City, China, 19–21 August 2009; pp. 324–329. [CrossRef]
- Muñoz-Nieto, A.; Rodríguez-Gonzálvez, P.; González-Aguilera, D.; Fernández-Hernández, J.; Gómez-Lahoz, J.; Picón-Cabrera, I.; Herrero-Pascual, J.; Hernández-López, D. UAV Archaeological reconstruction: The study case of Chamartin Hillfort (Avila, Spain). *ISPRS Ann. Photogramm. Remote. Sens. Spat. Inf. Sci.* 2014, *II-5*, 259–265. [CrossRef]
- 83. Traviglia, A.; Cowley, D.; Lambers, K. Finding common ground: Human and computer vision in archaeological prospection. *AARGnews- Newsl. Aer. Archaeol. Res. Group* **2016**, *53*, 14.
- Nicolas, T.; Gaugne, R.; Tavernier, C.; Millet, E.; Bernadet, R.; Gouranton, V. Lift the veil of the block samples from the Warcq chariot burial with 3D digital technologies. In Proceedings of the 2018 3rd Digital Heritage International Congress (DigitalHERITAGE) Held Jointly with 2018 24th International Conference on Virtual Systems & Multimedia (VSMM 2018), San Francisco, CA, USA, 26–30 October 2018; pp. 1–8. [CrossRef]
- 85. Ebinger-Rist, N.; Peek, C.; Stelzner, J.; Gauß, F. Computed Tomography: A Powerful Tool for Non-Destructive Mass Documentation of Archaeological Metals. In Proceedings of the Metal 2010 : Interim Meeting of the ICOM-CC Metal Working Group, Charleston, SC, USA, 11–15 October 2010; p. 458.
- Jungblut, D.; Karl, S.; Mara, H.; Krömker, S.; Wittum, G. Automated GPU-based surface morphology reconstruction of volume data for archaeology. In *Scientific Computing and Cultural Heritage*; Bock, H.G., Jäger, W., Winckler, M.J., Eds.; Springer: Berlin/Heidelberg, Germany, 2013; pp. 41–49. [CrossRef]
- 87. Payne, E.M. Imaging techniques in conservation. J. Conserv. Mus. Stud. 2013, 10, 17–29. [CrossRef]
- 88. Nicolas, T.; Gaugne, R.; Tavernier, C.; Arnaldi, V.G. La tomographie, l'impression 3D et la réalité virtuelle au service de l'archéologie. *Les Nouvelles de l'archéologie* 2016, 16–22. [CrossRef]
- 89. Franke, L.; Haehn, D. Modern scientific visualizations on the web. Informatics 2020, 7, 37. [CrossRef]
- 90. Shrivakshan, G.; Chandrasekar, C. A comparison of various edge detection techniques used in image processing. *Int. J. Comput. Sci. Issues (IJCSI)* **2012**, *9*, 269.
- 91. Peli, T.; Malah, D. A study of edge detection algorithms. Comput. Graph. Image Process. 1982, 20, 1–21. [CrossRef]
- 92. Krishnan, K.B.; Ranga, S.P.; Guptha, N. A survey on different edge detection techniques for image segmentation. *Indian J. Sci. Technol.* **2017**, *10*, 1–8. [CrossRef]
- 93. Turk, P.; Korat, L.; Miran, P. Ropotulje z Dežmanovih kolišč na Ljubljanskem barju: Prispevek k arheologiji zvoka/The rattles from Dežman's lake dwellings in the Ljubljana marshes: A contribution to the archaeology of sound. *Argo* **2018**, *61*, 10–17.
- 94. Hey, T.; Trefethen, A. The fourth paradigm 10 years on. *Inform. Spektrum* **2020**, *42*, 441–447. [CrossRef]
- Tolle, K.M.; Tansley, D.S.W.; Hey, A.J.G. The Fourth Paradigm: Data-Intensive Scientific Discovery [Point of View]. Proc. IEEE 2011, 99, 1334–1337. [CrossRef]
- 96. Agrawal, A.; Choudhary, A. Perspective: Materials informatics and big data: Realization of the "fourth paradigm" of science in materials science. *APL Mater.* **2016**, *4*, 053208. [CrossRef]
- 97. Hosny, A.; Parmar, C.; Quackenbush, J.; Schwartz, L.H.; Aerts, H.J. Artificial intelligence in radiology. *Nat. Rev. Cancer* 2018, 18, 500–510. [CrossRef]